

866 SOUTHEAST 3ND AVENUE SUITE 500 FORT LAUDERDALE, FLORIDA 33335-9002 TELEPHONE (954) 523-5685 TELEFAX (954) 760-9531 WORLD WIDE WEB http://www.adorna.com

October 30, 1996

DEPARTMENT OF STATE DIVISION OF CORPORATIONS 409 East Gaines Street Tallahassee, Florida 32399 600001994486---1 -11/01/96--01109--004 *****122.50 *****122.50

600001994436---1 -11/01/96--01103--005 ******8.75 ******8.75

98

1

PH 3:07

П

Re: Formation of Zenith AGC Acquisition Insurance Company

Dear Sir or Madam:

Enclosed please find the following items submitted on behalf of Zenith AGC Acquisition Insurance Company:

- 1. Articles of Incorporation to be filed to complete the formation of Zenith AGC Acquisition Insurance Company
- 2. Certificate of Designation of Registered Agent/Registered Office designating the State Treasurer and Insurance Commissioner as the registered agent pursuant to Section 624.422, Florida Statutes (original on file with the Florida Insurance Department)
- 3. Check in the amount of \$122.50 as filing fee for Articles of Incorporation, Designation of Registered Agent/Registered Office, and certified copy of Articles of Incorporation
- 4. Check in the amount of \$8.75 as filing fee for Certificate of Status

We would appreciate your expedited return of the certified copy of the Articles of Incorporation and Certificate of Status. Thank you for your attention to this matter.

Sincerely ori A. Lovgree

Enclosure

LAL/C.LTRGEN/19916

MIAM

20 11/4

ARTICLES OF INCORPORATION OF ZENITH AGC ACQUISITION INSURANCE COMPANY

The undersigned incorporators to these Articles of Incorporation, natural persons over the age of 18 years, competent to contract and the majority of whom are citizens of the United States of America, hereby form a stock insurer corporation under the laws of the State of Florida.

ARTICLE I. NAME

The name of the corporation shall be Zenith AGC Acquisition Insurance Company. The principal place of business of this corporation shall be 888 S.E. 3rd Avenue, Suite 500, Ft. Lauderdale, FL 33316, located in Broward County.

ARTICLE II. NATURE OF BUSINESS

The purpose of the corporation is to engage in every aspect of workers' compensation insurance.

ARTICLE III. CAPITAL STOCK

The maximum number of shares of stock that this corporation is authorized to issue is 10,000 shares of common stock having a par value of \$1.00 per share. The corporation shall not begin transacting business unless it achieves a capital and surplus of \$5 million.

ARTICLE IV. TERM OF EXISTENCE

The corporation shall exist perpetually.

ARTICLE V. INITIAL REGISTERED OFFICE AND AGENT

The initial registered office of this Corporation shall be 888 S.E. 3rd Avenue, Suite 500, Ft. Lauderdale, Florida, 33316, and the initial registered agent of this Corporation at such office shall be Michael Camilleri, who upon accepting this designation agrees to comply with the provisions of Section 48.091, Florida Statutes, as amended from time to time, with respect to keeping an office open to receive service of process from the Treasurer and Insurance Commissioner of the State of Florida.

ARTICLE VI. DIRECTORS

The corporation shall have five directors initially, all of whom are United States citizens and all of whom are over the age of 18. The names and residence street addresses of the directors whose initial terms of office shall be for one year are:

Stanley R. Zax 21255 Colifa Street Woodland Hills, CA 91367-5021

John J. Tickner 21255 Califa Street Woodland Hills, CA 91367-5021

James P. Ross 21255 Califa Street Woodland Hills, CA 91367-5021

Fredricka Taubitz 21255 Califa Street Woodland Hills, CA 91367-5021

Keith E. Trotman 21255 Califa Street Woodland Hills, CA 91367-5021

ARTICLE VII. INCORPORATORS

The names and residence street addresses of the incorporators, all of whom are over the age of 18 and all of whom are United States citizens are:

Stanley R. Zax 21255 Califa Street Woodland Hills, CA 91367-5021



James P. Ross 21255 Califa Street Woodland Hills, CA 91367-5021

Fredricka Taubitz 21255 Califa Street Woodland Hills, CA 91367-5021

Keith E. Trotman 21255 Califa Street Woodland Hills, CA 91367-5021

30th THE INCORPORATORS have hereunto set their hands and seals this _____ day of 19 96 October

STANLEY R.

TICKNER

aub FREDRICKA TAUBITZ

Keith E. Troman KEITH E. TROTMAN

ED NO THEASURER

OCT 22 1996

Legal Division

STATE OF CALIFORNIA COUNTY OF LOS ANGELES

PERSONALLY APPEARED before me, Stanley R. Zax, who acknowledged executing the above Articles of Incorporation.



Ruth Fulguin

State of Florida California At Large

My commission expires:

STATE OF CALIFORNIA COUNTY OF LOS ANGELES

PERSONALLY APPEARED before me, John J. Tickner, who acknowledged executing the above Articles of Incorporation.

3/3/97



Notary Public

State of Florida California At Large

My commission expires: 3/3/97

STATE OF CALIFORNIA COUNTY OF __ LOS ANGELES

PERSONALLY APPEARED before me, James P. Ross, who acknowledged executing the above Articles of Incorporation.

RUTH FULGIUM COMM. # 986569 Natary Public -- California LOS ANGELES COUNTY My Comm. Expires MAR 3, 1997 Cardin de Cardonidado

gun

Notary Public State of Ftorida California At Large

My commission expires: 3/3/97



STATE OF CALIFORNIA COUNTY OF LOS ANGELES

PERSONALLY APPEARED before me, Fredricka Taubitz, who acknowledged executing the above Articles of Incorporation.



Notary Public

State of Florida California At Large

My commission expires:

STATE OF CALIFORNIA COUNTY OF LOS ANGELES

PERSONALLY APPEARED before me, Keith E. Trotman, who acknowledged executing the above Articles of Incorporation.

3/3/97



My commission expires: ____3/3/97

Notary Public

State of Florida California At Large

ACCEPTANCE OF REGISTERED AGENT

Having been named to accept service of process from the Treasurer and Insurance Commissioner of the State of Florida for Zenith AGC Acquisition Insurance Company, at the place designated in the Articles of Incorporation, Michael Camilleri agrees to act in this capacity, and agrees to comply with the provisions of Section 48.091, Florida Statutes, relative to keeping open such office.

Date:

10/23/11

Signature

96

NON -1

PH

بب 80 FILED

LKK/C.HISC/15634

CONSENT ANT AGREEMENT IN RE SERVE TOF PROCESS

State of <u>California</u> } County of <u>Los Angeles</u> } ss.

Know All Men by These Presents, That the <u>Zenith AGC Acquisition Insurance Company</u> of _______,an insurer or other entity subject to the statutory agent for service of process provisions of the Florida Insurance Code duly organized and existing under and by virue of the Laws of <u>Florida</u>, does hereby agree and consent that actions may be commenced against it in any court having jurisidiction in any County in the State of Florida, in which a cause of action may arise, or in which the plaintiff may reside, by the service of process upon the Treasurer and Insurance Commissioner of the State of Florida, hereby stipulating and agreeing that such service shall be taken and held in all Courts to be as valid and binding upon this insurer or other entity as if personal service had been made upon the President or Secretary, or any other duly authorized and accredited officer thereof. The undersigned hereby further agrees and stipulates that this consent and agreement is and shall remain irrevocable, so long as there is flability, under any policy, claim or cause of action within this State, either fixed or contingent. Said insurer or other entity does hereby designate <u>Michael J. Camilleri</u>

as the name and address of the person to whom process against it served upon said Treasurer and Insurance Commissioner is to be forwarded. In the event of a change in the designation of the person to whom process is to be forwarded, the insurer or other entity shall immediately file a new Consent and Agreement form with the Insurance Commissioner.

In Witness Whereof, we, the President or Chief Executive Officer and Secretary of said insurer or other entity, have hereunto set our hands and affixed the seal of said insurer or other entity on this the _____ day of _____ day of ______ A.D. 19 96

SEAL

Zenith AGC Acquisition Insurance Company Bу President or Chief Executive Officer (Signature) Stapley R

- 1

: t *

Secretary (Signature) John J. Tickner

The foregoing agreement must be accompanied by a duly certified copy of the order or resolution of the Board of Directors of the company, association or society, or other entity regulated by Florida Insurance Code, authorizing the President and Secretary to execute the same for and on behalf of the corporation.

CONSENT OF PERSON TO WHOM PROCESS IS TO BE FORWARDED

The undersigned hereby consents and agrees to be the person to whom process against the above company, served upon the Treasurer and Insurance Commissioner, may be forwarded.

DI4-144 rev. 11/90

P9600090400

ARTICLES OF MERGER Merger Sheet

MERGING:

ħ

ZENITH AGC ACQUISITION INSURANCE COMPANY, #P96000090400, A FL CORP.

INTO

ZENITH INSURANCE COMPANY, a California corporation, F94000003596

File date: December 20, 1996, effective December 31, 1996 Corporate Specialist: Susan Payne

Division of Corporations - P.O. BOX 6327 - Tallahassee, Florida 32314

-+-

P96000090400

ADORNO & ZEDER

888 SOUTHEAST 3ND AVENUE SUITE 500 FORT LAUDERDALE, FLORIDA 33335-9002 TELEPHONE (954) 523-5885 TELEFAX (954) 760-9531 WORLD WIDE WEB http://www.sdorno.com

> Writers Direct (954) 713-1378

-12/31/96--01065--017

-01/13/97--01027--001 *****87.50 *****87.50

---:3

÷

****35.00

300002042333-

*****70.00

December 19, 1996

Secretary of State DIVISION OF CORPORATIONS 409 East Gaines Street Tallahassee, Florida 32399 Attn: Susan Payne

EFFECTIVE DATE

RE: Articles of Merger of Associated General Commerce Self Insurers' Fund and Zenith AGC Acquisition Insurance Company; and Articles of Merger of Zenith AGC Acquisition Insurance Company and Insurance Company.

Dear Ms. Payne:

C

Enclosed please find two (2) Original Articles of Merger, one for each of the above referenced companies. Also enclosed are two (2) checks, numbers 054371 and 054372, each in the amount of \$87.50 for payment of the filing fee and a certified copy of the filing.

Please forward the certified copies to my attention in the enclosed Federal Express package.

Thank you. Your anticipated prompt attention to this matter is greatly appreciated.

Sincerely. Charlene Bellinger Honig Enclosures FILING R AGENT CERT. COPY_ 25108/10819.002 CUS OVERFAYMENT 122.50 TOTAL

MIAM

NEW YORK

ARTICLES OF MERGER Merger Sheet

١.

T Vy

4,

.

MERGING:

,

ASSOCIATED GENERAL COMMERCE SELF INSURERS TRUST FUND, a group self-insurance fund authorized pursuant to 624.4621

INTO

ZENITH AGC ACQUISITION INSURANCE COMPANY, a Florida corporation, P96000090400

File date: December 20, 1996, effective December 31, 1996 Corporate Specialist: Susan Payne

Division of Corporations - P.O. BOX 6327 -Tallahassee, Florida 32314

ARTICLES OF MERGER OF ASSOCIATED GENERAL COMMERCE SELF INSURERS' TRUST FUND A Qualified Florida Self- Insurers' Fund

FILED SECRETARY OF STATE DIVISION OF CORPORATIONS 96 DEC 20 PM 2:44

EFFECTIVE DATE

12/31/96

10 a.m.

INTO

ZENITH AGC ACQUISITION INSURANCE COMPANY A Florida Corporation

Pursuant to FSA §§ 628.451 and 607.1105, the entities described herein, desiring to effect a merger, adopt the fole wing Articles of Merger:

ARTICLE I

The name of the corporation surviving the merger is ZENITH AGC ACQUISITION INSURANCE COMPANY. The name of the surviving corporation has not been changed as a result of the merger. The surviving corporation is a domestic corporation incorporated under the laws of the State of Florida.

ARTICLE II

The name of the nonsurviving entity is ASSOCIATED GENERAL COMMERCE SELF INSURERS' TRUST FUND. The nonsurviving entity is a group self-insurance fund authorized to operate pursuant to FSA §624.4621.

ARTICLE III

The Plan of Merger, containing the information required by FSA § 628.451 and § 607.1101 is set forth in Exhibit A, which is attached hereto and incorporated by reference herein. The Plan of Merger has been approved by the Florida Department of Insurance.

ARTICLE IV

The effective date of the Merger is 10:00 a.m., December 31, 1996.

ARTICLE V

The manner of adoption and vote of the surviving corporation, ZENITH AGC ACQUISITION INSURANCE COMPANY, was as follows:

The Marger was adopted by the Board of Directors of ZENITH AGC ACQUISITION INSURANCE COMPANY without shareholder action on November 5, 1996. A shareholder vote was not required.

ARTICLE VI

The manner of adoption and vote of the nonsurviving entity, ASSOCIATED GENERAL COMMERCE SELF INSURERS' TRUST FUND, was as follows:

The Merger was approved by the Board of Trustees of ASSOCIATED GENERAL COMMERCE SELF INSURERS' TRUST FUND on September 11, 1996. The Merger was approved by the members of ASSOCIATED GENERAL COMMERCE SELF INSURERS' TRUST FUND at a special meeting on November 13, 1996. The number of votes cast for the merger was sufficient for approval.

Signed this 18th day of December____, 1996.

ZENITH AGC ACQUISITION INSURANCE COMPANY Bv: 🔊 John JY Tickner, Secretary

ASSOCIATED GENERAL COMMERCE SELF INSURERS' TRUST FUND By: Davis & Oburburdt

Doris Oberhardt, Administrator

NPPROVEr NO TPEASUMER 062 13 1996

21880

121080

AGREEMENT AND PLAN OF METGER

BY AND AMONG

ZENITH AGC ACQUISITION INSURANCE COMPANY,

ZENITH INSURANCE COMPANY,

ZENITH NATIONAL INSURANCE CORP.,

ASSOCIATED GENERAL COMMERCE SELF-INSURERS' TRUST FUND

and

AGC RISK MANAGEMENT GROUP INC.

As of October 7, 1996

٦,

e.

TABLE OF CONTENTS

Page

ARTICLE I THE MERGER; EFFECTIVE TIME; CLOSING

1.1	The Merger	
1.2	Effective Time	2
1.3	Closing	2

ARTICLE II SURVIVING CORPORATION

2.1	Articles of Incorporation 2
2.2	By-Laws
2.3	Directors
2.4	Officers

ARTICLE III

MERGER CONSIDERATION

3.1	Right to Receive Merger Consideration	2
3.2	Determination of Adjusted GAAP Net Worth	3
3.3	Payment of Merger Consideration	
3.4	Assumption of Liability for Assessments	4

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF THE FUND

4.1	Organization and Qualification
4.2	Subsidiaries
4.3	Authority Relative to This Agreement 5
4.4	Consents and Approvals; No Violation
4.5	Financial Statements and Reports 6
4.6	Reserves
4.7	Absence of Certain Changes or Events 7
4.8	Litigation
4.9	No Regulatory Disqualifications 9
4.10	Proxy Statement
4.11	Taxes
4.12	Employee Benefit Plans; Labor Matters 10
4.13	Intentionally Omitted
4.14	Intellectual Property
4.15	Brokers and Finders 12
4.16	Fairness Opinion
4.17	Title to Property
4.18	Insurance

00777359.10-21511

i

٦,

4.19	No Default
4.20	Noncompliance with Laws
4.21	Management and Service Agreements
4.22	Investment Portfolio and Other Assets 14
4.23	Intercompany and Affiliate Transactions; Insider Interests 14
4.24	Officers, Directors, Trustees and Employees 14
4.25	No Material Adverse Change 14
4.26	No Undisclosed Linbilities
4.27	Full Disclosure

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF ZENITH, ZENITH INSURANCE AND ZENITH NATIONAL

5.1	Corporate Organization and Qualification
5.2	Authority Relative to This Agreement
5.3	Consents and Approvals; No Violation 16
5.4	Proxy Statement
5.5	Brokers and Finders

•

ARTICLE VI

ADDITIONAL COVENANTS AND AGREEMENTS

6.1	Conduct of Business of the Fund
6.2	Alternative Proposals
6.3	Members' Approval; Proxy Statement 19
6.4	Satisfaction of Conditions, Receipt of
	Necessary Approvals
6.5	Access to Information
6.6	Publicity
6.7	Indemnification of Trustees and Officers
6.8	Guaranty by Zenith Insurance and Zenith
	National
6.9	Merger of Zenith and Zenith Insurance 21
6.10	Termination of RMG Management
	Agreements
6.11	Dividend Payments 22
6.12	Financial Statements and Reports 22

ARTICLE VII

CONDITIONS PRECEDENT TO CONSUMMATION OF THE MERGER

7.1	Conditions to Each Party's Obligations to Effect the Merger	22
7.2	Additional Conditions to the Obligations of Zenith	23
	Additional Conditions to the Obligations of the Fund	

0077359.10-21514

.

ARTICLE VIII TERMINATION

8.1	Termination by Mutual Consent
8.2	Termination by Either Zenith or the Fund
8.3	Termination by Zenith
8.4	Termination by the Fund
8.5	Effect of Termination

.

ARTICLE IX

MISCELLANEOUS AND GENERAL

9.1	Payment of Expenses and Other Payments	27
9.2	Survival of Representations and Warranties	28
9.3	Modification or Amendment	28
9.4	Waiver and Extension	28
9.5	Counterparts	28
9.6	Governing Law	28
9.7	Notices	28
9.8	Entire Agreement; Assignment	30
9.9	Parties in Interest	30
9.10	Certain Definitions	30
9.11	Validity	35
9.12	Captions	35
9.13	Further Assurances	36

.

..

...

.

.

.

•

,

٠.

EXHIBITS

.

.

.

.-.

~

.

Exhibit 7.2(c)	Form of Opinion of Gerald S. Livingston, Esq., Counsel to the Fund
Exhibit 7.2(f)	Form of Consulting Agreement
Exhibit 7.2(g)	Form of Transfer Agreement
Exhibit 7.3(d)	Form of Opinion of Counsel to Zenith
Exhibit 7.3(f)	Form of Management Agreement

0077359.10-2151a

iv

•

•

٦

AGRI:EMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER (this "Agreement"), dated as of October 7, 1996, by and among ZENITH AGC ACQUISITION INSURANCE COMPANY, a Florida corporation (together with its successors and assigns, "Zenith"), ZENITH INSURANCE COMPANY, a California corporation ("Zenith Insurance"), ZENITH NATIONAL INSURANCE CORP., a Delaware corporation ("Zenith National"), ASSOCIATED GENERAL COMMERCE SELF-INSURERS' TRUST FUND, a qualified Florida self-insurers fund f/k/a Associated General contractors Self-Insurers' Trust Fund (the "Fund"), and AGC RISK MANAGEMENT GROUP INC., a Florida not-for-profit corporation ("RMG").

RECITALS

WHEREAS, the respective Boards of Directors of Zenith, Zenith Insurance and Zenith National, and the Board of Trustees of the Fund have, subject to the conditions of this Agreement, determined that the Merger (as defined in Section 1.1) and the other transactions contemplated by this Agreement are in the best interests of their stockholders and policyholders, respectively, and have approved this Agreement and the transactions contemplated hereby; and

WHEREAS, Zenith, Zenith Insurance and Zenith National, on one hand, and the Fund, on the other hand, require (each from the other) certain representations, warranties, covenants and agreements in connection with the Merger.

NOW, THEREFORE, in consideration of the foregoing and the mutual representations, warranties, covenants and agreements set forth herein, Zenith, Zenith Insurance, Zenith National, the Fund and RMG hereby agree as follows:

ARTICLE I

THE MERGER; EFFECTIVE TIME; CLOSING

1.1 The Merger. Subject to the terms and conditions of this Agreement, at the Effective Time, the Fund shall be merged with and into Zenith and the separate existence of the Fund shall thereupon cease (the "Merger"). Zenith shall be the successor or surviving corporation in the Merger and shall continue to be governed by the Laws of the State of Florida under the name "Zenith AGC Acquisition Insurance Company," and the separate corporate existence of Zenith with all its rights, privileges, immunities, powers and franchises shall continue unaffected by the Merger. The corporation surviving the Merger is sometimes hereinafter referred to as the "Surviving Corporation." The Merger shall have the effects set forth herein and in the FBCA and the FIL. From and after the Effective Time, the Surviving Corporation shall possess all of the assets and other rights, privileges, immunities, powers of Zenith and the Fund, and shall be liable for all of the liabilities of each of Zenith and the Fund.

٦,

1.2 Effective Time. As soon as practicable after the satisfaction or waiver of the conditions to the Merger set forth in this Agreement, Zenith and the Fund will cause appropriate articles of merger (the "Articles of Merger") to be executed and filed with the Secretary of State of the State of Florida, as provided in the FBCA. Subject to the approval of the Articles of Merger by the Secretary of State of the State of Florida, the Merger will become effective as of 10:00 a.m. on December 31, 1996 or, if later, the date and time of the filing of the Articles of Merger (the "Effective Time").

1.3 <u>Closing</u>. The Fund shall as promptly as practicable notify Zenith, and Zenith shall as promptly as practicable notify the Fund, when the conditions to such party's or parties' obligation to effect the Merger contained in Section 7.1 have been satisfied or waived. The Closing shall take place (a) at the offices of RMG, 3504 Lake Lynda Drive, Suite 400, Orlando, Florida, at 10:00 a.m., Florida time, on the Closing Date or (b) at such other place, time and date as Zenith and the Fund may agree.

ARTICLE II

SURVIVING CORPORATION

2.1 <u>Articles of Incorporation</u>. The Articles of Incorporation of Zenith shall be the Articles of Incorporation of the Surviving Corporation until thereafter amended as provided by Law and the Articles of Incorporation.

2.2 <u>By-Laws</u>. The By-Laws of Zenith, as in effect immediately prior to the Effective Time, shall be the By-Laws of the Surviving Corporation until thereafter amended as provided by Law, the Articles of Incorporation and the By-Laws.

2.3 <u>Directors</u>. The directors of Zenith at the Effective Time shall, from and after the Effective Time, be the directors of the Surviving Corporation until their successors have been duly elected or appointed and qualified or until their carlier death, resignation or removal in accordance with the Articles of Incorporation and Surviving Corporation's By-laws.

2.4 <u>Officers</u>. The officers of Zenith at the Effective Time shall, from and after the Effective Time, be the initial officers of the Surviving Corporation until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Articles of Incorporation and the By-Laws.

ARTICLE III

MERGER CONSIDERATION

3.1 Right to Receive Merger Consideration.

ŝ

As of the Effective Time, each Member Distributee shall, by virtue of the Merger and without any action on the part of such Member Distributee, be entitled to receive, in accordance

0077359.10-2151=

with the Plan of Distribution, its pro rata portion (based on premiums paid during the five-year period ending on the Effective Date) of the cash payments to be made pursuant to Section 3.3 (collectively, the "Merger Consideration").

3.2 Determination of Adjusted GAAP Net Worth.

(a) Prior to the Closing Date, Zenith Insurance shall retain a "big six" accounting firm selected by the Management Group (the "Accountant") to perform the audits and prepare the reports required pursuant to this Section 3.2; provided, however, that if the Management Group fails to select an accounting firm within 3 business days prior to the Closing Date, Zenith Insurance may select the accounting firm that will act as the Accountant.

(b) Within sixty (60) days after the Effective Date, the Accountant shall conduct and complete an audit of the Fund's statement of financial position as of the Effective Date, which shall be prepared in accordance with GAAP without giving effect to the Merger (the "Closing Balance Sheet"). The Accountant shall deliver its audit report on the Closing Balance Sheet to Zenith, Zenith Insurance, Zenith Insurance's auditors and the Management Group. RMG shall cooperate with the Accountant in connection with such audit, including the provision of customary representation letters. The Accountant's audit report on the Closing Balance Sheet shall be final and binding on all of the parties hereto.

(c) From time to time after the Effective Date, but prior to each payment required pursuant to Section 3.3 (b), (c) and (d), the Accountant shall audit a revised statement of financial position of the Fund as of the Effective Date, prepared in accordance with GAAP without giving effect to the Merger (each, a "Revised Closing Balance Sheet"). The Accountant shall issue, and deliver to Zenith, Zenith Insurance, Zenith Insurance's auditors and the Management Group, a special report on each Revised Closing Balance Sheet, in accordance with the American Institute of Certified Public Accountants' Statements on Auditing Standards. Each of the Accountant's special reports shall be final and binding on all of the parties hereto.

(d) "Adjusted GAAP Net Worth" shall mean, as of any date, (i) the amount reflected as "member distribution payable" on the most recently prepared Revised Closing Balance Sheet, plus (ii) \$1,600,000, minus (iii) 50% of the Accountant's fees and expenses for auditing the Closing Balance Sheet and any Revised Closing Balance Sheet that has been prepared, minus (iv) payments made to Members after the Effective Date in respect of Loss Sensitive Policies to the extent that the amount reflected as "member distribution payable" on the most recently prepared Revised Closing Balance Sheet represents an accrual for such payments.

3.3 <u>Payment of Merger Consideration</u>. Zenith shall.pay the Merger Consideration to the Member Distributees, as follows:

(a) As soon as practical after the Effective Time, Zenith shall make payments to Member Distributees equal, in the aggregate, to \$1,140,000;

(b) On or before December 31, 1997, Zenith shall make payments to Member Distributees equal, in the aggregate, to 50% of the amount by which (i) the Fund's Adjusted GAAP Net Worth, as of the date of payment, net of any amounts included therein that represent accruals for

٦

payments due to Members under Loss Sensitive Policies, exceeds (ii) the amount paid pursuant to paragraph (a) hereof;

(c) On or before December 31, 1998, Zenith shall make payments to Member Distributees equal, in the aggregate, to 50% of the amount by which (i) the Fund's Adjusted GAAP Net Worth, as of the date of payment, net of any amounts included therein that represent accruals for payments due to Members under Loss Sensitive Policies, exceeds (ii) the sum of the amounts paid pursuant to paragraphs (a) and (b) hereof; and

(d) On or before December 31, 1999, Zenith shall make payments to Member Distributees equal, in the aggregate, to the sum of:

(i) 100% of the amount by which the Fund's Adjusted GAAP Net Worth, as of the date of payment, exceeds the sum of the amounts paid pursuant to paragraphs (a), (b) and (c) hereof;

(ii) interest accrued at the rate of 7% per annum on each of the amounts paid pursuant to the foregoing paragraphs (a), (b) and (c) and clause (d)(i) hereof, in each case, from the Effective Date until the respective date of payment of such amounts pursuant to this Section 3.3 (each, a "Payment Date");

(iii) interest accrued at the rate of 7% per annum on the interest calculated pursuant to the foregoing clause (ii) from the respective Payment Date until the date of payment pursuant to this clause (d).

3.4 <u>Assumption of Liability for Assessments</u>. From and after the Effective Time, Zenith hereby assumes all Liability for Assessments and Zenith shall indemnify and hold harmless the Members against any Liability for Assessments.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF THE FUND AND RMG

Except as described in the appropriate sections of the Disclosure Schedule, the Fund and RMG represent and warrant to Zenith. Zenith Insurance and Zenith National that:

4.1 Organization and Qualification.

(a) The Fund is a self-insurance trust fund duly organized, validly existing and in good standing under the Laws of the State of Florida. RMG is a not-for profit corporation duly organized, validly existing and in good standing under the Laws of the State of Florida. Each of the Fund and RMG has all requisite ; ower and authority and all necessary governmental Consents to own, lease and operate its properties and to carry on its business as it is now being conducted. The Fund and RMG have heretofore made available to Zenith complete and correct copies of the Fund's Declaration of Trust and RMG's Articles of Incorporation and By-Laws, respectively, each in effect as of the date hereof.

(b) The Fund is (i) duly licensed or authorized as a self-insurance trust fund in the State of Florida, (ii) not required to be licensed or authorized as a self-insurance trust fund or insurance company in any jurisdiction other than the State of Florida, and (iii) duly authorized in the State of Florida to write the line of business reported as being written in the SAP Statements.

(c) RMG is (i) duly licensed or authorized as a managing general agent and was previously licensed as a service company in the State of Florida, and (ii) not required to be licensed or authorized as a managing general agent or service company in any jurisdiction other than the State of Florida.

4.2 <u>Subsidiaries</u>. The Fund has no Subsidiaries. Except for RMG, the Fund does not directly or indirectly have any equity or similar interest in, or any interest convertible into or exchangeable or exercisable for any equity or similar interest in, any corporation, partnership, joint venture or other business association or entity that directly or indirectly conducts any activity which is material to the Fund. RMG has one member, which member is the Fund.

4.3 Authority Relative to This Agreement. The Fund and RMG have the requisite power and authority to execute and deliver this Agreement and, subject to approval of this Agreement by the Requisite Percentage of Voting Members in accordance with the FIL as contemplated by Section 6.3 of this Agreement, to consummate the transactions contemplated hereby. This Agreement and the consummation by the Fund and RMG of the transactions contemplated hereby have been duly and validly authorized by the Board of Trustees of the Fund and the Board of Directors of RMG, respectively, and no other proceedings on the part of the Fund or RMG are necessary to authorize this Agreement or to consummate the transactions contemplated hereby (other than the approval of this Agreement by the Requisite Percentage of Voting Members in accordance with the FIL). This Agreement has been duly and validly executed and delivered by the Fund and RMG and, assuming this Agreement constitutes the valid and binding agreement of Zenith, Zenith Insurance and Zenith National, constitutes the valid and binding agreement of the Fund and RMG, enforceable against the Fund and RMG in accordance with its terms, except that the enforcement hereof may be limited by (a) bankruptcy, insolvency, reorganization, moratorium or other similar Laws now or hereafter in effect relating to creditors' rights generally and (b) general principles of equity (regardless of whether enforceability is considered in a proceeding in equity or at law).

4.4 <u>Consents and Approvals: No Violation</u>. Neither the execution, delivery or performance of this Agreement by the Fund or RMG nor the consummation by the Fund or RMG of the transactions contemplated hereby nor compliance by the Fund or RMG with any of the provisions hereof will (a) conflict with or result in any breach of any provision of the Declaration of Trust of the Fund, or the Articles of Incorporation or By-Laws of RMG; (b) require any Consent of any governmental or regulatory authority, except (i) the filing of the Articles of Merger pursuant to the FBCA and appropriate documents with other relevant authorities, (ii) the filing of appropriate documents with, and approval of, the Florida Commissioner; (c) result in a Default under any of the terms, conditions or provisions of any Contract to which the Fund or RMG may be a party, or Permit by which the Fund or RMG or any of their assets may be bound, except for such Defaults as to which requisite waivers or consents have been obtained; or (d) assuming the Consents and Permits referred to in this Section 4.4 are duly and timely obtained or made and the approval of this Agreement by the Requisite Percentage of Voting Members has been obtained, violate any Order or Law applicable to the Fund or RMG or any of their assets.

4.5 Financial Statements and Reports.

(a) <u>SAP Statements</u>. The Fund has delivered to Zenith complete and correct copies of all SAP Statements relating to periods ending, or prepared as of any date, on or after December 31, 1993. Each SAP Statement (and the exhibits and schedules relating thereto) was prepared in accordance with SAP applied on a consistent basis (except for changes, if any, disclosed therein) and is complete in all material respects, and each SAP Statement fairly presents (in accordance with SAP) the financial position of the Fund, as of the respective dates thereof, or its earnings and changes in surplus or cash flows, as the case may be, for and during the respective periods covered thereby. There were no material liabilities affecting the Fund as of December 31, 1995 required in accordance with SAP to be reflected or disclosed in the SAP Statements for the period then ended, or as of June 30, 1996 required in accordance with SAP to be reflected or disclosed in the SAP Statements for the period then ended, which are not so reflected or disclosed.

(b) GAAP Statements.

(i) The Fund has delivered to Zenith the Interim Statements and the GAAP Statements. The GAAP Statements and Interim Statements (including the notes thereto) have been prepared in accordance with GAAP applied on a consistent basis throughout the period indicated and present fairly the financial position of the Fund, as of the respective dates thereof, and the results of its operations and cash flows for the periods then ended, subject, in the case of the Interim Statements, to normal year-end adjustments which will not, in the aggregate, in the opinion of the Trustees of the Fund, be material.

(ii) RMG has delivered to Zenith its audited financial statements for the fiscal year ended September 30, 1995, together with the notes related thereto. Such financial statements (including the notes thereto) have been prepared in accordance with GAAP applied on a consistent basis throughout the period indicated and present fairly the financial position of RMG as of the respective dates thereof and the results of its operations and cash flows for the periods then ended.

(c) The Fund has filed all SAP Statements, together with all exhibits and schedules thereto, required to be filed with or submitted to the appropriate regulatory authorities of the State of Florida on forms prescribed or permitted by such authority. The Fund has filed all other required forms, reports and documents with the Florida Commissioner required to be filed or submitted by it pursuant to the FIL and the rules and regulations thereunder, all of which have complied in all material respects with all applicable requirements of such laws and the rules promulgated thereunder. None of such forms, reports or documents, including, without limitation, any financial statements or schedules included therein at the time filed, contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

4.6 Reserves.

(a) Section 4.6 of the Disclosure Schedule sets forth the methodology used by the Fund in developing the reserves for incurred losses, incurred loss adjustment expenses, incurred but not reported losses and loss adjustment expenses for incurred but not reported losses as set forth in the SAP Statements. All such reserves in respect of insurance or reinsurance treaties or agreements established or reflected in the SAP Statements were determined in accordance with commonly accepted actuarial standards applied on a consistent basis and are in compliance, in all material respects, with the requirements of the FIL, as well as those of any other applicable insurance laws.

(b) All reserves and accrued liabilities for estimated losses, settlements, costs and expenses from pending suits, actions and proceedings included in each SAP Statement were determined in accordance with SAP and generally accepted actuarial assumptions and Statement of Financial Accounting Standards No. 5 issued by the Financial Accounting Standards Board.

4.7 <u>Absence of Certain Changes or Events</u>. Except as specifically provided for in this Agreement and except as set forth in Section 4.7 of the Disclosure Schedule, since December 31, 1995, the Fund and RMG have operated their respective businesses only in the ordinary course of business and consistent with past practice, and except in the ordinary course of business and consistent with past practice, neither the Fund nor RMG has:

(i) declared, set aside or paid any dividends or declared or made any other distributions of any kind to its policyholders;

(ii) incurred any obligation or liability (fixed or contingent), or series of related obligations or liabilities, other than ordinary course obligations arising pursuant to the terms of its insurance or reinsurance treaties and contracts;

(iii) made any change in its accounting or reserving methods or practices, other than those required to be made by regulatory agencies or by the Financial Accounting Standards Board, or made any change in depreciation or amortization policies or rates adopted by it which change materially affects any financial statement item (without giving effect to any offsetting change):

(iv) made any change in the underwriting, establishment of reserves, investment or claims adjustment policies and practices or change in any activity which (A) has had the effect of accelerating the recording and billing of premiums or accounts receivable or delaying the payment of expenses or establishing loss reserves in connection with any accounts or business of the Fund, or (B) has had the effect of materially altering, modifying or changing the historic, financial or accounting practices or policies of the Fund, including accruals of, and reserves for, tax liabilities;

(v) amended its Declaration of Trust or By-laws or merged with or into or consolidated with any other person, or changed or agreed to change in any manner the character of its business;

۲

(vi) suffered any material damage, destruction or loss, whether or not covered by insurance;

(vii) entered into any agreement, commitment or transaction (including, without limitation, any capital expenditure, capital financing or sale of assets) for any material amount, other than ordinary course obligations pursuant to the terms of its insurance or reinsurance treaties and contracts;

(viii) allowed any Lien on any tangible or intangible asset, or any sale, transfer, assignment or lease of any tangible or intangible asset;

(ix) effected any cancellation of any debt or waiver or release of any material right or claim other than under its insurance or reinsurance treaties and contracts;

(x) made any payment, discharge or satisfaction of any claim, liability or obligation (absolute, accrued, contingent or otherwise), other than ordinary course payments pursuant to the terms of its insurance or reinsurance treaties and contracts;

(xi) been subjected to any labor dispute, litigation or governmental investigation material to their business or their financial condition;

(xii) made any loan or advance to any person;

(xiii) purchased, leased or acquired any assets or properties of any person;

(xiv) increased or agreed to increase any salary, wages, benefits or other forms of compensation payable or to become payable to any of its current or former officers, directors, trustees or employees, or any bonus or severance payments or arrangements made to, for or with any of its officers, directors, trustees or employees, or any supplemental retirement plan or other program or special remuneration or compensation for any of its officers, directors, trustees or employees, other than those which have been approved by Zenith;

(xv) terminated or amended or failed to perform its obligations or caused the occurrence of any default under any agreement the result of which had, or in the future can reasonably be expected to have, a Material Adverse Effect;

(xvi) entered into any agreement, whether in writing or otherwise, to do any of the foregoing; or

(xvii) entered into any material transaction or commitment of any kind other than in the ordinary course of business.

4.8 <u>Litigation</u>. Except as disclosed in the financial statements delivered pursuant to Section 4.5 or as set forth in Section 4.8 of the Disclosure Schedule, there is no Litigation pending or, to the knowledge of the Fund, threatened against or affecting the business operations or financial condition of the Fund or RMG. Neither the Fund nor RMG is in default with respect to any judgement, order, writ, injunction or decree of any court or any Federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality.

4.9 <u>No Regulatory Disqualifications</u>. To the knowledge of the Fund, no event has occurred and no condition exists or, to the extent it is within the reasonable control of the Fund, will occur or exist with respect to the Fund or RMG that, in connection with obtaining any regulatory Consents required for the Merger, would cause the Fund or RMG to fail to satisfy on its face any applicable statute or written regulation of any applicable insurance regulatory authority, which is reasonably likely to adversely affect the Fund's or RMG's ability to consummate the transactions contemplated hereby.

4.10 <u>Proxy Statement</u>. The Proxy Statement will comply with the FIL, except that no representation is made by the Fund with respect to information supplied in writing by Zenith or any affiliate of Zenith specifically for inclusion in the Proxy Statement. None of the information as to the Fund or RMG contained in the Proxy Statement will contain, and none of the information heretofore furnished to Zenith or its representatives by the Fund contains, any untrue statement of a material fact required to be stated therein or omits to state a material fact necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

4.11 <u>Taxes</u>. Except as disclosed in Section 4.11 of the Disclosure Schedule (with paragraph reference corresponding to those set forth below) or in the SAP Statements or the GAAP Statements or as disclosed in writing to Zenith prior to the date hereof:

(a) All Tax Returns required to be filed with respect to the Fund and RMG have been duly and timely filed, and all such Tax Returns are true, correct and complete. All Taxes that are due and payable by the Fund and RMG have been paid or adequate provision has been made on the Fund's and RMG's books and records in accordance with GAAP or SAP, as the case may be, for the payment of such Taxes. There are no Liens on any of the assets of the Fund e...cept for statutory Liens for current Taxes.

(b) With respect to any period for which Tax Returns have not yet been filed, or for which Taxes are not yet due or owing, the Fund or RMG, as the case may be, has made due and sufficient current accruals for such Taxes in accordance with SAP and GAAP, and such current accruals are duly and fully provided for in the GAAP Statements of the Fund or RMG, as the case may be.

(c) The Fund and RMG have complied (and until the Effective Time will comply) with all applicable Laws relating to information reporting and the withholding of Taxes in connection with amounts paid or owing to any third party.

(d) Neither the Fund nor RMG have been audited or examined by the Internal Revenue Service. Except as set forth in Section 4.11(d) of the Disclosure Schedule, the Fund and RMG have not been required to file any state, local or foreign income or franchise Tax Returns. There are no outstanding agreements, waivers or arrangements extending the statutory period of limitations applicable to any claim for, or the period for the collection or assessment of. Taxes due from the Fund or RMG for any taxable period. The Fund has delivered to Zenith copies, which are true, correct and complete, of each of the federal income and state premium Tax Returns, for each of the last three taxable years, filed by the Fund and RMG.

(c) No audit or other proceeding by any court, governmental or regulatory authority, or similar person is pending or to the knowledge of the Fund, threatened with respect to any Taxes due from the Fund or RMG or any Tax Return filed by or relating to the Fund or RMG. No assessment of Tax has been proposed, orally or in writing, against the Fund or RMG.

(f) Neither the Fund nor RMG is a party to, nor is either of them bound by, nor does either have any obligation under, any tax sharing contract, and neither the Fund nor RMG has any liability for indemnification of third parties with respect to Taxes or liabilities for Taxes as a transferee.

(g) For all tax years for which the applicable statute of limitations has not yet expired, the Fund's and RMG's respective tax reserves have been computed in accordance with the requirements of the Code.

4.12 Employee Benefit Plans; Labor Matters.

(a) <u>General Compliance with Law</u>. Each Plan has been operated in accordance with its terms and the requirements of ERISA, the Code, and all other applicable Laws. All reports and disclosures relating to the Plans required to be filed or furnished to Δy governmental entity, participants or beneficiaries prior to the Closing have been filed in a timely manner and in accordance in all material respects with applicable Law, except as set forth in Section 4.12(a) of the Disclosure Schedule. Section 4.12 of the Disclosure Schedule contains a true and complete list of each Plan and ERISA Affiliate Plan.

ERISA Title IV Liability: Defined Benefit Plans. (i) Neither the Fund nor **(b)** RMG, nor any ERISA Affiliate of the Fund or RMG has incurred any direct or indirect liability under, arising out of, or by operation of Title IV of ERISA that has not been satisfied in full, and no fact or event exists that could reasonably be expected to give rise to any such liability, other than liability for premiums due the PBGC (which premiuns have been paid when due): (ii) for each Plan and ERISA Affiliate Plan which is subject to Title IV of ERISA, the aggregate accumulated benefit obligation (as determined under Statement of Financial Accounting Standards No. 87) of such Plan or ERISA Affiliate Plan does not exceed the fair market value of the assets of such Plan or ERISA Affiliate Plan; (iii) no Plan or any trust established thereunder that is subject to Section 302 of ERISA and Section 412 of the Code has incurred any "accumulated funding deficiency" (as defined in Section 302 of ERISA and Section 412 of the Code), whether or not waived; (iv) all contributions required to be made with respect thereto (whether pursuant to the terms of any Plan or otherwise) have been timely made; (v) no Lien exists under Section 412(n) of the Code or Section 4068 of ERISA with respect to any assets of the Fund or RMG; (vi) no tax under Section 4971 of the Code has been incurred with respect to any Plan; and (:ii) the Fund and RMG and each ERISA Affiliate of the Fund and RMG does not sponsor, maintain, contribute to, or is required to contribute (and has not within the past six (6) years contributed) to a "multiemployer pension plan," as defined in Section 3(37) of ERISA, or a plan described in Section 4063(a) of ERISA.

(c) <u>Prohibited Transactions: Fiduciary Duties</u>. (i) Neither the Fund, RMG, nor any Plan, nor any trust created thereunder and any trustee or administrator thereof has engaged in a transaction in connection with which the Fund or any ERISA Affiliate, any Plan, any such trust, or any trustee or administrator thereof, or any party dealing with any Plan or any such trust, which could result in a civil penalty assessed pursuant to Section 409 or 502(i) of ERISA or a tax imposed pursuant to Section 4975 of the Code; and (ii) the Fund and all fiduciaries (as defined in Section 3(21) of ERISA) with respect to the Plans, have complied in all respects with Section 404 of ERISA.

(d) <u>Determination Letters</u>. (i) Each Plan intended to be qualified under Section 401(a) of the Code has received a favorable determination letter from the Internal Revenue Service with respect to the Tax Reform Act of 1986 and other applicable Laws, or an application was filed for such determination letter on a timely basis, and (ii) nothing has occurred from the date of such letter or such filing that could reasonably be expected to affect the qualified status of such Plan.

(c) <u>No Acceleration of Liability</u>. The consummation of the transactions contemplated by this Agreement will not (i) entitle any current or former employee, trustee, director or officer of the Fund or RMG to severance pay, unemployment compensation or any other payment, except as expressly provided in this Agreement or (ii) accelerate the time of payment or vesting, or increase the amount of compensation or benefit due any such employee, trustee, director or officer.

(f) <u>Ability to Terminate Plans: Post-Retirement Plans</u>. Each Plan is terminable in accordance with the terms expressly set forth therein, except as may be limited by applicable Law. No Plan provides death, medical or other benefits with respect to current or former employees after retirement or other termination of service, other than coverage mandated by applicable Law.

Employee Relations. Except to the extent set forth in Section 4.12(g) of (g) the Disclosure Schedule, (i) there is no labor strike, dispute, slowdown, stoppage or lockout actually pending or threatened against or affecting the Fund or RMG and during the past three years there has not been any such action; (ii) no union claims to represent the employees of the Fund or RMG; (iii) neither the Fund nor RMG is a party to or bound by any collective bargaining or similar agreement with any labor organization, or work rules or practices agreed to with any labor organization or employee association applicable to employees of the Fund or RMG; (iv) none of the employees of the Fund or RMG are represented by any labor organization and the Fund and RMG do not have any knowledge of any current union organizing activities among the employees of the Fund or RMG, nor does any question concerning representation exist concerning such employees; (v) true and correct copies of all written personnel policies, rules and procedures applicable to employees of the Fund or RMG have heretofore been delivered to Zenith by the Fund and RMG; (vi) there is no grievance arising out of any collective bargaining agreement or other grievance procedure pending against the Fund or RMG; (vii) neither the Fund nor RMG has received notice of the intent of any federal, state, local or foreign agency responsible for the enforcement of labor or employment laws to conduct an investigation of the Fund or RMG, nor is such an investigation in progress; (viii) there are no employment contracts or severance agreements with any employees of the Fund or RMG; and (ix) no employee of the Fund or RMG has suffered an "employment loss" (as defined in the Worker Adjustment and Retraining Notification Act) during the three (3) month period prior to the Closing.

4.13 Intentionally Omitted.

0077359.10-21514

4.14 <u>Intellectual Property</u>. (a) Each of the Fund and RMG is the owner of, or a licensee under a valid license for, all items of intellectual property which are material to its business as currently conducted, including, without limitation, (i) copyrights, patents, trademarks, logos, service marks, trade names, service names, all applications therefor und all registrations thereof, and (ii) rights and licenses, computer software, trade secrets, know-how, inventions, processes, formulae and other intellectual property rights (collectively, the "Intellectual Property"); (b) with respect to all Intellectual Property owned by the Fund or RMG, the Fund or kMG, as the case may be, is the sole owner and has the exclusive right to use such Intellectual Property, and such owned Intellectual Property is not subject to any Liens; (c) there is no infringement or other adverse claim against the rights of the Fund or RMG with respect to any of the Intellectual Property; and (d) neither the Fund nor RMG has not been charged with, nor to the knowledge of the Fund or RMG, is the Fund or RMG threatened to be charged with nor is there any basis for any such charge of, infringement or other violation of, nor has the Fund or RMG infringed, nor is it infringing upon, any unexpired rights of any third party in any of the Intellectual Property.

4.15 <u>Brokers and Finders</u>. Neither the Fund nor RMG has employed any investment banker, broker, finder, consultant or intermediary in connection with the transactions contemplated by this Agreement which would be entitled to any investment banking, brokerage, finder's or similar fee or commission in connection with this Agreement or the transactions contemplated hereby.

4.16 <u>Fairness Opinion</u>. The Fund has engaged Raymond James, as financial advisor to the Fund, to render a fairness opinion with respect to the consideration to be received by the Members in the Merger. Upon receipt of such opinion, a copy thereof will be provided to Zenith.

4.17 Title to Property.

(a) Each of the Fund and RMG (i) has good, valid and marketable title to all of its properties, assets and other rights that do not constitute real property, free and clear of all Liens, and (ii) owns, or has valid leasehold interests in or valid contractual rights to use, all of the assets, tangible and intangible, used by, or necessary for the conduct of, its business.

(b) Each of the Fund and RMG:

(i) is in peaceful and undisturbed possession of the space and/or estate under each real property lease under which it is a tenant, and there are no defaults by it as tenant thereunder; and

(ii) has good and valid rights of ingress and egress to and from all the real property leased by it from and to the public street systems for all usual street, road and utility purposes.

4.18 <u>Insurance</u>. Section 4.18 of the Disclosure Schedule is a complete and accurate list of all primary, excess and umbrella policies of general liability, fire, products liability, employers' liability, directors' and officers' liability, workers' compensation, bonds and any other form of insurance owned or held by or on behalf of or providing insurance coverage to the Fund and RMG, including the following information for each such policy: type(s) of insurance coverage provided; name of insurer; effective dates of the policy; policy number; per occurrence and annual aggregate deductibles or self-insured retention; and per occurrence and annual aggregate limits of liability; and the extent, if any, to which the limits of liability of any products liability or general liability insurance have been invaded or exhausted. The Fund and RMG are, and have been continuously since January 1, 1995, insured with financially responsible insurers in such amounts and against such risks and losses as are customary for companies conducting the business as conducted by the Fund and RMG during such time period. Neither the Fund nor RMG is in Default under, nor have they received any notice of cancellation or termination with respect to, any insurance policies. The Fund and RMG maintain all insurance coverage required to be maintained pursuant to the FIL, inclu-ing, without limitation, excess insurance in accordance with Florida Administration Code Rule 4-190.061.

4.19 <u>No Default</u>. Neither the Fund nor RMG is in Default under any term, condition or provision of (a) its Declaration of Trust, or Articles of Incorporation, respectively, (b) any Contract or other instrument or obligation to which it is a party or by which it or any of its properties or assets may be bound or affected; (c) any Order applicable to it or any of its properties or assets; or (d) any Permit necessary for it to conduct its business as currently conducted.

4.20 <u>Noncompliance with Laws</u>. The businesses of the Fund and RMG are being conducted in compliance with all applicable Laws. Except as set forth in Section 4.20 of the Disclosure Schedule, since January 1, 1993, neither the Fund nor RMG has received any written notification or written communication from any agency or department of federal, state, or local government (a) asserting that the Fund or RMG is not in compliance with any of the Laws, Orders or Permits of any governmental agency or authority or that any such agency or authority enforces, or (b) requiring the Fund or RMG to enter into or consent to the issuance of a cease and desist order, formal agreement, directive or commitment which restricts the conduct of its business or which affects its capital, its credit or reserve policies, its management, or the payment of dividends.

4.21 Management and Service Agreements.

(a) All management and/or service agreements entered into by the Fund as now in force are set forth in Section 4.21 of the Disclosure Schedule, and to the extent required under applicable Law, are in a form acceptable to applicable regulatory authorities or, to the extent required by Law, have been filed and not objected to by such authorities within the period provided for objection.

(b) Except as set forth in Section 4.21 of the Disclosure Schedule, each agreement set forth in Section 4.21 of the Disclosure Schedule is valid and binding against the Fund and each other party thereto, and is in full force and effect in accordance with its terms. Except as set forth in Section 4.21 of the Disclosure Schedule, the Fund and each other party thereto is and has been in compliance with the terms of such agreements. There are no defaults (or circumstances or events that, with the giving of notice or lapse of time or both, would become defaults) with respect to any such contract or other agreement and no such contract or other agreement contains any provisions providing that the other party thereto may terminate the same by reason of the transactions contemplated by this Agreement or any other provision which would be altered or otherwise become applicable by reason of such transactions. 4.22 <u>Investment Portfolio and Other Assets</u>. Section 4.22 of the Disclosure Schedule contains the Fund's investment guidelines and a list of the securities and other investments in the Fund's investment portfolio, as of June 30, 1996, with information included thereon as to the amortized cost of each such investment and the market value thereof as of such date. None of such investments included in the investment portfolio is in default in the payment of principal or interest or dividends or materially impaired. All such investments included in such portfolio comply with all insurance laws and regulations of each of the states to which the Fund is subject relating thereto including, without limitation, Florida Administrative Code Rule 4-190.071.

4.23 Intercompany and Affiliate Transactions; Insider Interests. Section 4.23 of the Disclosure Schedule lists all intercompany agreements or arrangements of any kind between or among the Fund, RMG or their officers, directors, trustees or policyholders.

4.24 Officers, Directors, Trustees and Employees. The Fund and RMG have previously delivered to Zenith true and correct information as to (a) the name of each of the Fund's officers, directors, trustees and employees, (b) the name of each of RMG's officers, directors and employees as of September 30, 1996, and (c) the total compensation of all such officers, directors, employees and trustees.

4.25 <u>No Material Adverse Change</u>. Since December 31, 1995, there has been no change in the business, properties, assets, condition (financial or otherwise), liabilities or operations of the Fund or RMG which, individually or in the aggregate has had, or is reasonably likely to have, a Material Adverse Effect. The Fund is not aware of any fact or facts which, individually or in the aggregate, is or are reasonably likely to have a Material Adverse Effect.

4.26 No Undisclosed Liabilities. Except for liabilities and obligations (x) incurred in the ordinary course of business and consistent with past practice or (y) pursuant to the terms of this Agreement, since December 31, 1995 to the date hereof, neither the Fund nor RMG has incurred any liabilities or obligations of any nature, whether or not accrued, contingent or otherwise, that have, or would be reasonably likely to have, individually or in the aggregate a Material Adverse Effect or would be required by GAAP or SAP to be reflected on a balance sheet of the Fund or RMG (including the notes thereto).

4.27 <u>Full Disclosure</u>. The representations and warranties made by the Fund in this Agreement, or in any documents referenced or delivered pursuant hereto or thereto, do not, and will not, contain any untrue statement of a material fact or omit to state a material fact required to be stated herein or therein, or necessary to make the statements and facts contained herein or therein, in light of the circumstances in which they are made, not false or misleading. Copies of all documents heretofore or hereafter delivered or made available to Zenith pursuant hereto were or will be complete and accurate copies of such documents.

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF ZENITH, ZENITH INSURANCE AND ZENITH NATIONAL

Each of Zenith, Zenith Insurance and Zenith National represents and warrants to the

Fund that:

5.1 Corporate Organization and Qualification.

(a) Zenith, Zenith Insurance and Zenith National are corporations duly organized, validly existing and in good standing under the Laws of the States of Florida, California and Delaware, respectively, and each is qualified and in good standing as a foreign corporation in each jurisdiction where the properties owned, leased or operated, or the business conducted, by it require such qualification. Zenith is a wholly owned subsidiary of Zenith Insurance, and Zenith Insurance is a wholly owned subsidiary of Zenith National. Zenith, Zenith Insurance, Zenith National, and each Subsidiary of Zenith Insurance has all requisite corporate power and authority and all necessary governmental Consents to own, lease and operate its properties and to carry on its business as it is now being conducted.

(b) Zenith Insurance is (i) duly licensed or authorized as an insurance company in its jurisdiction of incorporation, (ii) duly licensed or authorized as an insurance company in each other jurisdiction where it is required to be so licensed or authorized, and (iii) duly authorized in its jurisdiction of incorporation and each other applicable jurisdiction to write each line of business reported as being written in Zenith's Annual Statement for the year ended December 31, 1995. Zenith, Zenith Insurance and Zenith National have heretofore provided to the Fund complete and correct copies of their charters and bylaws, each as in effect as of the date hereof.

(c) Prior to the Effective Time, Zenith will be duly licensed or authorized as an insurance company in the State of Florida and duly authorized in the State of Florida to write workers' compensation insurance business.

5.2 <u>Authority Relative to This Agreement</u>. Each of Zenith, Zenith Insurance and Zenith National has the requisite corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. This Agreement and the consummation by Zenith, Zenith Insurance and Zenith National of the transactions contemplated hereby have been duly and validly authorized by the Boards of Directors of Zenith, Zenith Insurance and Zenith National, and no other corporate proceedings on the part of Zenith, Zenith Insurance or Zenith National are necessary to authorize this Agreement or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Zenith, Zenith Insurance and Zenith National and, assuming this Agreement constitutes the valid and binding agreement of the Fund and RMG, constitutes the valid and binding agreement of Zenith, Zenith Insurance and Zenith National, enforceable against Zenith, Zenith Insurance and Zenith National in accordance with its terms, except that the enforcement bereof may be limited by (a) bankruptcy, insolvency, reorganization, moratorium or other similar Laws now or hereafter in effect relating to creditors' rights generally and (b) general principles of equity (regardless of whether enforceability is considered in a proceeding at law or in equity).

0077359.10-21514

5.3 Consents and Approvals: No Violation. Neither the execution, delivery or performance of this Agreement by Zenith or Zenith Insurance nor the consummation by Zenith or Zenith Insurance of the transactions contemplated hereby nor compliance by Zenith or Zenith Insurance with any of the provisions hereof will (a) conflict with or result in any breach of any provision of the Articles of Incorporation or By-Laws of Zenith or Zenith Insurance; (b) require any Consent of any governmental or regulatory authority, except (i) in connection with the applicable requirements of the HSR Act, (ii) the filing of the Articles of Merger pursuant to the laws of the State of Florida, the filing of an agreement of merger pursuant to the laws of the State of California, and the filing of appropriate documents with the relevant authorities of other states in which Zenith insurance or any of its Subsidiaries is authorized to do business, (iii) the filing of appropriate documents with, or approval of, the respective Commissioners of Insurance of the states of Florida, California and other states in which Zenith Insurance conducts business; (c) result in a Default under any of the terms, conditions or provisions of any Contract to which Zenith, Zenith Insurance or any of its Subsidiaries or any of their respective assets may be bound, except for such Defaults as to which requisite waivers or consents have been obtained; or (d) assuming the Consents referred to in this Section 5.3 are duly and timely obtained or made, violate any Order or Law applicable to Zenith, Zenith Insurance or any of its Subsidiaries or to any of their respective assets.

5.4 <u>Proxy Statement</u>. None of the information supplied by Zenith or its affiliates in writing specifically for inclusion in the Proxy Statement will, at the time the Proxy Statement is mailed, at the time of the Special Meeting, or at the Effective Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

5.5 <u>Brokers and Finders</u>. Zenith and its affiliates have not employed any investment banker, broker, finder, or intermediary in connection with the transactions contemplated by this Agreement which would be entitled to any investment banking, brokerage, finder's or similar fee or commission in connection with this Agreement or the transactions contemplated hereby.

ARTICLE VI

ADDITIONAL COVENANTS AND AGREEMENTS

6.1 <u>Conduct of Business of the Fund</u>. During the period from the date of this Agreement to the Effective Time (unless Zenith shall otherwise agree in writing and except as otherwise contemplated by this Agreement), the Fund will conduct its operations according to its ordinary and usual course of business consistent with past practice and shall use all reasonable efforts to preserve intact its current business organization, keep available the services of its current officers and employees, maintain its Permits and Contracts and preserve its relationships with customers, suppliers and others having business dealings with it. Without limiting the generality of the loregoing, and except as otherwise contemplated by this Agreement, the Fund will not, without the prior written consent of Zenith:

(i) adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of the Fund (other than the Merger);

(ii) except as set forth in Section 6.1(ii) of the Disclosure Schedule, adopt any amendments to its Declaration of Trust;

(iii) make any acquisition, by means of merger, consolidation or otherwise, or enter into any agreement for the disposition, of assets or securities;

(iv) other than in the ordinary course of business consistent with past practice, incur any indebtedness for borrowed money or guarantee any such indebtedness or make any loans, advances or capital contributions to, or investments in any other person other than the Fund;

(v) grant any increases in the compensation of any of its directors or trustees or, except in the ordinary course of business and in accordance with past practice, grant any increases in the compensation of any of its officers, employees or agents;

(vi) except in connection with the transactions contemplated by this Agreement, enter into any new or amend any existing employment agreement or, except as may be consistent with Fund policies in effect as of the date of this Agreement, enter into any new or amend any existing severance or termination agreement with any officer or employee of the Fund;

(vii) except as may be required to comply with applicable Law, become obligated under any new pension plan, welfare plan, multiemployer plan, employee benefit plan, severance plan or similar plan, which was not in existence on the date hereof, or amend any Plan;

(viii) amend, increase, accelerate the payment or vesting of the amount payable or to become payable under or fail to make any required contribution to, any benefit plan or increase any non-salary benefits payable to any employee or former employee, except in the ordinary course of business consistent with past practice;

(ix) change any method of accounting or accounting practice by the Fund, except for any such required change in GAAP or applicable statutory accounting principles;

(x) change its investment guidelines or policies or conduct transactions in investments except in compliance with its investment guidelines and policies and approved programs or transactions and all applicable insurance Laws;

(xi) enter into any Contract to purchase, or to lease for a term in excess of one year, any real property, provided that the Fund may as a tenant renew any existing lease for a term not to exceed two years;

(xii) enter into any insurance, reinsurance, coinsurance or similar Contract, whether as insurer, reinsurer or reinsured, except in the ordinary course of business consistent with past practice;

(xiii) enter into any Contract with any insurance agent or broker that provides, by its terms, for exclusivity (including, without limitation, by territory, product, or distribution) or that is not terminable by its terms within 180 days by the Fund without substantial premium or penalty or, in the case of career agents, without commission renewal liability, except to the extent that the Contract provides for vesting commissions;

(xiv) (x) take, or agree or commit to take any action that would make any representation and warranty of the Fund hereunder inaccurate at the Effective Time (except for representations and warranties which speak as of a particular date, which need be accurate only as of such date), (y) omit, or agree or commit to omit, to take any action necessary to prevent any such representation or warranty from being inaccurate at the Effective Time (except for representations and warranties which speak as of a particular date, which need be accurate only as of such date), or (z) take, or agree or commit to take, any action that would result in, or is reasonably likely to result in, any of the conditions of the Merger set forth in Article VII not being satisfied;

(xv) authorize, recommend, propose or announce an intention to do any of the foregoing, or enter into any contract, agreement, commitment or arrangement to do any of the foregoing; or

(xvi) settle any tax audit, or make or change any tax election or file amended Tax Returns.

6.2 <u>Alternative Proposals</u>. The Fund will not authorize, and will use its best efforts to cause its officers, trustees, directors, employees or agents not to, directly or indirectly, solicit, initiate or encourage any inquiries relating to, or the making of any proposal which constitutes, an Alternative Proposal, or recommend or endorse any Alternative Proposal, or participate in any discussions or negotiations, or provide third parties with any nonpublic information, relating to any such inquiry or proposal or otherwise facilitate any effort or attempt to make or implement an Alternative Proposal; provided, however, that the Fund may, and may authorize and permit its officers, trustees, directors, employees or agents to provide third parties with nonpublic information reasonably necessary to facilitate an Alternative Proposal, recommend or endorse any Alternative Proposal with or by any third party, and participate in discussions and negotiations with any third party relating to any Alternative Proposal, if the Fund's Board of Trustees, after having consulted with and considered the advice of outside counsel, has reasonably determined in good faith that the failure to do so would be inconsistent with its fiduciary duties to the Fund's Members under applicable Law. If the Fund enters into a definitive agreement with respect to any Alternative Proposal, it shall concurrently with entering into such agreement pay, or cause to be paid to Zenith the expenses outlined in Section 9.1(b). The Fund will immediately cease and cause to be terminated any activities, discussions or negotiations conducted prior to the date of this Agreement with any parties other than Zenith with respect to any of the foregoing. The Fund shall immediately advise Zenith following the receipt by it of any Alternative Proposal and the material terms and conditions thereof, and the identity of the person making any such Alternative Proposal, and advise Zenith of any developments with respect to such Alternative Proposal immediately upon the occurrence thereof.

6.3 Members' Approval; Proxy Statement. The Fund shall duly call, give notice of, convene and hold the Members' Meeting as soon as practicable following the execution of this Agreement for the purpose of obtaining the approval of the Requisite Percentage of the Voting Members with respect to this Agreement and the Merger. The Fund shall use its best efforts to obtain and furnish the information required to be included by it in the Proxy Statement to be furnished to the Fund's Members regarding the Merger. Zenith shall have the right to approve the form and contents of the Proxy Statement and any preliminary versions thereof prior to the time it is filed with the Florida Commissioner. After consultation with Zenith, the Fund shall respond promptly to any comments made by the Florida Commissioner with respect to the Proxy Statement and any preliminary version thereof, and cause the Proxy Statement to be mailed to its Voting Members, at the earliest practicable time following the execution of this Agreement. The Fund shall use its best efforts to obtain the approval of the Kaquisite Percentage of the Voting Members with respect to this Agreement and the Merger, and the Fund shall, through its Board of Trustees, subject to its fiduciary duties under applicable Law, recommend to its Members approval of this Agreement, unless, in each case, the members of the Board of Trustees of the Fund, after having consulted with and considered the advice of outside counsel, reasonably determine in good faith that under the circumstances the foregoing actions would be reasonably likely to result in a breach of their fiduciary duties to the Members under applicable Law. Notwithstanding the foregoing, the Board of Trustees of the Fund may at any time prior to the Effective Time withdraw, modify, or change any recommendation and declaration regarding this Agreement, or recommend and declare advisable any other offer or proposal, if the Board of Trustees, after consultation with its outside counsel, has reasonably determined in good faith that the making of such recommendation, or the failure to withdraw, modify or change its recommendation, would be reasonably likely to result in a breach of their fiduciary duties to the Members under applicable Law. Additionally, the Board of Trustees of the Fund may withdraw, modify or change any recommendation or declaration regarding this Agreement if the fairness opinion rendered by Raymond James in connection with this Agreement and the Related Documents determines the value of the consideration to be received by the Members of the Fund not to be fair to the Members, from a financial point of view.

6.4 <u>Satisfaction of Conditions, Receipt of Necessary Approvals</u>. Subject to the terms and conditions herein provided, each of the parties hereto agrees to (i) promptly, effect all necessary registrations, submissions and filings, including, but not limited to, filings under the HSR Act and submissions of information requested by governmental authorities, which may be necessary or required in connection with the consummation of the transactions contemplated by this Agreement, (ii) to use its best efforts to secure federal and state antitrust clearance (including taking steps to avoid or set aside any preliminary or permanent injunction or other order of any federal or state court of competent jurisdiction or other governmental authority), (iii) use its best efforts to take all other action and to do all other things necessary, proper or advisable to consummate and make effective as promptly as practicable the transactions contemplated by this Agreement and to obtain all other necessary or appropriate Consents (including but not limited to any required Consents of the Commissioners). Each of the parties hereto acknowledges that certain actions may be necessary with respect to the foregoing in making notifications and obtaining Consents which are material to the consummation of the transactions contemplated hereby, and each of the parties hereto agrees to take such action as is necessary to complete such notifications and obtain such Consents,

0077359.10-21514
provided, however, that nothing in this Section 6.4 or elsewhere in this Agreement shall require any party hereto to hold separate or make any divestiture of any asset or otherwise agree to any restriction on their operations which would in any such case be material to the assets, liabilities or business of, (a) in the case of the Fund or RMG, the Fund, and (b) in the case of Zenith, Zenith Insurance or Zenith National, Zenith National and its Subsidiaries, taken as a whole, in order to obtain any Consent required by this Agreement. All necessary registrations, submissions and filings under the HSR Act or in connection with securing federal or state antitrust clearance, shall be at the cost of Zenith, provided, however, that the HSR Act filing fee shall be shared equally by Zenith and the Fund. Additionally, Zenith shall be primarily responsible for all filings required hereunder, unless such filings are specifically required of the Fund or RMG, in which case the Fund or RMG, as the case may be, shall promptly make such filings following receipt of written notification of the requirement for such filings.

6.5 Access to Information.

(a) Upon reasonable notice, the Fund shall afford to officers, employees, counsel, accountants and other authorized representatives of Zenith ("Representatives"), in order to evaluate the transactions contemplated by this Agreement, reasonable access, during normal business hours throughout the period prior to the Effective Time, to its properties, books and records and, during such period, shall furnish promptly to such Representatives all information concerning its business, properties and personnel as may reasonably be requested.

(b) Zenith agrees that it will not, and will cause its Representatives not to, use any information obtained pursuant to this Section 6.5 for any purpose unrelated to the consummation of the transactions contemplated by this Agreement.

(c) Zenith will hold, and will use its best efforts to cause its Representatives to hold, in strictest confidence, unless compelled to disclose by judicial or administrative process, or, in the opinion of its counsel, by other requirements of Law, all documents and information concerning the Fund furnished to Zenith or its Representatives in connection with the transactions contemplated by this Agreement (except to the extent that such information can be shown to have been (i) previously known by Zenith or its Representatives, (ii) in the public domain through no fault of Zenith or its Representatives, or (iii) later lawfully acquired by Zenith or its Representatives from other sources (unless Zenith knows that such other sources are not entitled to disclose such information)) and will not release or disclose such information to any other person, except its Representatives in connection with this Agreement, provided that such Representative shall have first been advised by Zenith of the confidentiality provision of this Section 6.5.

(d) Notwithstanding the provisions hereof, during the period prior to the Effective Time, the parties shall take appropriate precautions to ensure that competitively sensitive information is not exchanged in a manner which is inconsistent with applicable Law.

6.6 <u>Publicity</u>. Zenith and the Fund will consult with each other and will mutually agree upon any press releases or public announcements pertaining to the Merger and shall not issue any such press releases or make any such public announcements prior to such consultation and agreement, except as may be required by applicable Law or by obligations pursuant to any listing agreement with any national securities exchange, in which case the party proposing to issue such press

release or make such public announcement shall use its best efforts to consult in good faith with the other party before issuing any such press releases or making any such public announcements.

6.7 Indemnification of Trustees and Officers. Zenith shall indemnify and hold harmless the Indemnified Parties from and against all costs, damages, judgments, attorney's fees, expenses, obligations and liabilities of whatsoever kind or nature, without limitation as to time, amount or otherwise, which the Indemnified Parties may incur or sustain in connection with any claim, action, suit, proceeding or investigation arising out of or pertaining to any action or omission in their capacity as trustee, fiduciary, officer, employee or agent of the Fund or RMG (including, without limitation, any which arise out of or relate to actions or omissions which relate to the Merger and the transactions contemplated by this Agreement); provided, however, that Zenith shall not be obligated to indemnify the Indemnified Parties if and to the extent that such costs, damages, judgments, attorney's fees, expenses, obligations or liabilities arise as a result of (a) a violation of criminal law resulting in a conviction, (b) any deliberately dishonest or fraudulent act or omission, or (c) any act or omission committed in bad faith or with malicious purpose or constituting gross negligence, and, in the case of any claim, action, suit, proceeding or investigation arising out of or pertaining to the Merger or any transaction contemplated by this Agreement or the Related Documents, Zenith shall not be obligated to indemnify the Indemnified Parties if and to the extent that all material aspects of the Murger and the transactions contemplated by this Agreement and the Related Documents were not fully described and disclosed in the Proxy Statement. At its option, Zenith may provide insurance policies of its choosing which insure and cover Zenith's obligation of indemnification to the Indemnified Parties as contained herein; provided, however, that the provision of such insurance polices by Zenith shall not relieve Zenith from its obligation of indemnification to the Indemnified Parties, and provided further, that any deductibles required under any such insurance policies shall be the sole responsibility of Zenith.

6.8 <u>Guaranty by Zenith Insurance and Zenith National</u>. Zenith Insurance and Zenith National hereby jointly and severally guarantee the performance by Zenith of its covenants and obligations under this Agreement and the Related Documents. The parties acknowledge and agree that this guaranty by Zenith Insurance and Zenith National constitutes an important inducement to the Fund to execute this Agreement and, consequently, constitutes an essential part of the consideration to be received by the Fund hereunder. This guaranty by Zenith National shall survive the merger of Zenith and Zenith Insurance, and shall continue in full force until all obligations of Zenith (and all obligations of Zenith Insurance acquired pursuant to the merger of Zenith and Zenith Insurance) under this Agreement and the Related Documents have been satisfied and discharged. Zenith Insurance and Zenith National shall not be discharged from liability under this guaranty so long as any obligation of Zenith existing under this Agreement or the Related Documents remains outstanding and unperformed. This guaranty by Zenith Insurance and Zenith National shall inure to the benefit of the Fund, its Members, the Indemnified Parties and the parties to the Related Documents.

6.9 Merger of Zenith and Zenith Insurance. As soon as practical after the Effective Time, Zenith, Zenith Insurance and Zenith National shall cause Zenith to be merged with and into Zenith Insurance, with (a) Zenith Insurance as the surviving corporation in such merger, (b) the outstanding shares of Zenith being cancelled and no shares of Zenith Insurance being issued in exchange therefor, and (c) the outstanding shares of Zenith Insurance remaining outstanding and unaffected by such merger, all in accordance with the GCL and the FBCA. The merger of Zenith and Zenith Insurance shall have the effect set forth in the GCL, including (i) the separate existence of Zenith shall cease, (ii) Zenith Insurance shall succeed, without other transfer, to all the rights and property of Zenith, and (iii) Zenith Insurance shall be subject to all the debts and liabilities of Zenith in the same manner as if Zenith Insurance had itself incurred them.

6.10 <u>Termination of RMG Management Agreements</u>. The Fund and RMG hereby agree that, from and after the Effective Time, all management agreements between RMG and the Fund are hereby terminated and of no further force or effect.

6.11 <u>Dividend Payments</u>. As soon as practical after the Effective Time, Zenith shall, subject to the approval of the Florida Department, pay dividends on the Fund's Loss Sensitive Policies equal to \$1,381,583.

6.12 Financial Statements and Reports.

(a) As promptly as practicable after the date hereof, but in no event later than five business days prior to the Closing Date, the Fund and RMG shall prepare and deliver to Zenith financial statements of the Fund that present fairly the financial position of the Fund as of September 30, 1996, and the results of its operations and cash flows for the period then ended in accordance with GAAP (the "September Financial Statements").

(b) As promptly as practicable after the date hereof, but in no event later than five business days prior to the Closing Date, the Fund and RMG shall cause Milliman & Robertson, Inc. to prepare and deliver to Zenith an actuarial review of the Fund's reserves as of September 30, 1996.

(c) As promptly as practicable after the Fund or RMG files or submits any financial statement or other report to the insurance regulatory authorities of the State of Florida (but in no event later than five business days thereafter), the Fund and RMG shall deliver to Zenith a true and correct copy of each such statement and report, together with all exhibits and schedules thereto.

(d) After the date hereof, the Fund and RMG shall continue to prepare in the ordinary course and shall deliver, as soon as available, to Zenith, true and complete copies of such financial statements (including, but not limited to, annual and quarterly financial statements prepared in accordance with GAAP and SAP), reports or analyses as may be prepared or received by the Fund or RMG and as relate to any of the business, operations or affairs of the Fund or RMG, including, without limitation, normal internal reports which the Fund or RMG prepares and special reports (such as those of financial consultants).

ARTICLE VII

CONDITIONS PRECEDENT TO CONSUMMATION OF THE MERGER

7.1 <u>Conditions to Each Party's Obligations to Effect the Merger</u>. The respective obligations of each party to effect the Merger are subject to the satisfaction or, where permissible, waiver at or prior to the Closing Date of the following conditions (provided that any such condition may not be waived without the consent of the Florida Commissioner):

(a) <u>Member Approval</u>. This Agreement shall have been duly approved by the Requisite Percentage of the Voting Members in accordance with applicable Law.

(b) <u>Fairness Opinion</u>. Raymond James shall have delivered to the Fund and Zenith its written opinion that the transactions contemplated by this Agreement and the Related Documents are fair, from a financial point of view, to the Members.

(c) <u>No Injunctions</u>. There shall not be in effect any Law or Order of a court or governmental or regulatory agency of competent jurisdiction directing that the transactions contemplated herein not be consummated; <u>provided</u>, <u>however</u>, that, subject to the terms and provisions herein provided, prior to invoking this condition each party shall use its reasonable efforts to have any such Order vacated.

(d) <u>Governmental Filings and Consents</u>. Subject to the terms and provisions herein provided all governmental Consents legally required for the consummation of the Merger and the transactions contemplated hereby shall have been obtained and be in effect on the Closing Date (including but not limited to the approval of the Commissioners and any Consents which may be required under the insurance Laws of any state in which Zenith Insurance conducts any business or owns any assets), without any limitations unacceptable to Zenith, and the waiting periods under the HSR Act shall have expired or been terminated.

7.2 <u>Additional Conditions to the Obligations of Zenith</u>. The obligation of Zenith to effect the Merger is subject to the satisfaction at or prior to the Closing Date of the following conditions, any or all of which may be waived in whole or in part by Zenith, to the extent permitted by applicable Law:

(a) <u>Representations and Warranties</u>. For purposes of this Section 7.2(a), the accuracy of the representations and warranties of the Fund and RMG set forth in Article IV of this Agreement shall be assessed as of the date of this Agreement and as of the Closing Date with the same effect as though all such representations and warranties had been made on and as of the Closing Date (provided that representations and warranties which are confined to a specified date shall speak only as of such date). All representations and warranties set forth in Article IV which are qualified by reference to materiality shall be true and correct as qualified and all other representations and warranties set forth in Article IV of this Agreement shall be true and correct in all material respects.

(b) <u>Performance</u>. Each of the Fund and RMG shall have performed, in all material respects, all covenants and agreements herein required to be performed by it prior to the Closing Date.

(c) <u>Officer's Certificate</u>. Zenith shall have received on the Closing Date a certificate, dated the Closing Date and executed by the Chairman of the Board of Trustees of the Fund, certifying as to the fulfillment of the conditions specified in Sections 7.2(a) and (b) hereof.

(d) <u>Documents</u>. Zenith and its counsel shall have received all such documents and instruments, or copies thereof, certified, if requested, as may be reasonably requested in connection with the compliance with the conditions precedent to Zenith's obligations hereunder. (c) <u>Opinion</u>. Zenith shall have received an opinion prepared in accordance with the Florida Opinion Standards established by the business section of the Florida Bar Association, dated the Closing Date, from Gerald S. Livingston, Esq., counsel to the Fund, in form and substance satisfactory to it and its counsel and substantially in the form of Exhibit 7.2(e) hereto.

(f) <u>Consulting Agreement</u>. The Management Group shall have entered into an agreement with Zenith, substantially in the form annexed hereto as Exhibit 7.2(f) (the "Consulting Agreement"), pursuant to which the Management Group shall, for five years after the Effective Date, provide consulting services to Zenith in connection with insurance business written by the Fund prior to the Effective Date and the development of Zenith's workers' compensation business in Florida, as required by Zenith as a condition to entering into this Agreement. The Consulting Agreement shall provide, among other thing₃, u.at (i) each member of the Management Group shall be compensated by Zenith for his or her service₃ in the amount of \$25,000 per year, and (ii) the Management Group (other than Doris Oberhardt) shall not engage in or undertake any workers compensation insurance business beginning on the date of the Effective Time and ending July 31, 2006.

(g) <u>Transfer Agreement</u>. RMG and ZRM shall have entered into an agreement, substantially in the form annexed hereto as Exhibit 7.2(g) (the "Transfer Agreement"), pursuant to which (i) RMG shall transfer all of its assets to ZRM, and ZRM shall assume all of the liabilities of RMG, (ii) ZRM may seek to hire any employees of RMG that ZRM deems to be necessary or useful in the operation or management of ZRM's business, and (iii) as soon as practical after the Effective Time, RMG shall take all appropriate action to effect its dissolution and liquidation in accordance with Florida law.

(h) <u>Accountant's Letter</u>. Shores & Company, P.A. shall have delivered to Zenith a letter, dated the Closing Date, in form and substance reasonably satisfactory to Zenith, containing statements and information of the type customarily included in accountants "comfort letters" with respect to the September Financial Statements, including, without limitation, the following:

(i) Shores & Company, P.A. are independent with respect to the Fund;

(ii) with respect to the nine-month period ended September 30, 1996, they have (A) performed the procedures specified by the American Institute of Certified Public Accountants for a review of the interim financial information as described in SAS No. 71, "Interim Financial Information," on the September Financial Statements, and (B) inquired of certain officials of the Fund and RMG who have responsibility for financial and accounting matters; and

(iii) nothing came to their attention as a result of the foregoing procedures that caused them to believe that (A) any material modifications should be made to the September Financial Statements for them to be in conformity with generally accepted accounting principles, or (B) there has been any decrease in the Fund's "member distribution payable" as compared with the amount shown in the September Financial Statements. (i) <u>Terminction of Pension Plan</u>. The Fund shall have taken all appropriate action to effect the termination of the Associated General Contractors Self-Insurers' Fund Pension Plan in accordance with applicable Law, and Zenith shall have received evidence reasonably satisfactory to it of such action.

(j) <u>Opinion Regarding Guaranty Fund Liability</u>. Zenith shall have received an opinion, dated the Closing Date, from counsel to Zenith, in form and substance satisfactory to Zenith, to the effect that from and after the Effective Time, none of Zenith National, Zenith or Zenith Insurance shall have any liability under the Florida Self-Insurance Fund Guaranty Association Act, FIL Section 631.90 et. seq. ("FIGA"), except for liability of the Fund under FIGA with respect to adjustments, if any, to the assessment of the Fund for the 1996 calendar year.

7.3 <u>Additional Conditions to the Obligations of the Fund</u>. The obligation of the Fund to effect the Merger is subject to the satisfaction at or prior to the Closing Date of the following conditions, any and all of which may be waived in whole or in part by the Fund to the extent permitted by applicable Law:

(a) <u>Representations and Warranties</u>. For purposes of this Section 7.3(a), the accuracy of the representations and warranties set forth in Article V of this Agreement shall be assessed as of the date of this Agreemen. and as of the Closing Date with the same effect as though all such representations and warranties had been made on and as of the Closing Date (provided that representations and warranties which are confined to a specified date shall speak only as of such date). All representations and warranties set forth in Article V of this Agreement which are qualified by reference to materiality shall be true and correct and all other representations and warranties set forth in Article V of this Agreement shall be true and correct in all material respects.

(b) <u>Performance</u>. Zenith, Zenith Insurance and Zenith National shall have performed, in all material respects, all covenants and agreements herein that are required to be performed by Zenith, Zenith Insurance and Zenith National prior to the Closing Date.

(c) <u>Officer's Certificate</u>. The Fund shall have received at the Closing Date a certificate dated the Closing Date and executed by the Chief Executive Officer or the Chief Financial Officer of Zenith certifying to the fulfillment of the conditions specified in Sections 7.3(a) and (b) hereof.

(d) <u>Opinion</u>. The Fund shall have received an opinion, dated the Closing Date, from counsel to Zenith, in form and substance satisfactory to it and its counsel and substantially in the form of Exhibit 7.3(d) hereto.

(e) <u>Formation of ZRM</u>. Zenith Insurance shall have formed ZRM and appointed each of the members of the Management Group as a director of ZRM.

(f) <u>Management Agreement</u>. Zenith shall have entered into an agreement with ZRM, substantially in the form of Exhibit 7.3(f) hereto (the "Management Agreement"), pursuant to which ZRM shall provide management services to Zenith after the Merger.

ARTICLE VIII

TERMINATION

8.1 <u>Termination by Mutual Consent</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Closing Date, before or after the approval by Members of the Fund, by the mutual written consent of Zenith and the Fund.

8.2 <u>Termination by Either Zenith or the Fund</u>. This Agreement may be terminated and the Merger may be abandoned by Zenith or the Fund, before or after the approval by Members of the Fund, if (i) any court of competent jurisdiction in the United States or some other governmental body or regulatory authority shall have issued an Order permanently restraining, enjoining or otherwise prohibiting the Merger and such Order shall have become final and nonappealable, <u>provided</u>, that the party seeking to terminate this Agreement pursuant to this clause (i) shall have used all reasonable efforts to remove such Order, (ii) the Merger shall not have been consummated by December 31, 1996; <u>provided</u> that the right to terminate this Agreement pursuant to this clause (ii) shall not be available to any party whose failure to fulfill any of its obligations under this Agreement shall have been voted on by Voting Members of the Fund and the vote shall not have been sufficient to satisfy the condition set forth in Section 7.1(a).

8.3 <u>Termination by Zenith</u>. This Agreement may be terminated by Zenith and the Merger may be abandoned prior to the Closing Date, before or after the approval by Members of the Fund, (i) in the event of a breach by the Fund or RMG of any covenant or agreement contained in this Agreement which, by its nature, cannot be cured prior to the Closing or which has not been cured within 30 days after the giving of written notice to the Fund of such breach, (ii) in the event of an inaccuracy of any representation or warranty of the Fund or RMG contained in this Agreement which, by its nature, cannot be cured prior to the Closing . which has not been cured within 30 days after the giving of written notice to the Fund of such inaccuracy and which inaccuracy, in either case, would cause the conditions set forth in Section 7.2(a) not to be satisfied, (iii) in the event that any of the conditions precedent to the obligations of Zenith to consummate the Merger cannot be satisfied or fulfilled by the date set forth in Section 8.2(ii) of this Agreement, provided that the failure of such conditions to be so satisfied shall not be as a result of Zenith's failure to fulfill its obligations under this Agreement, or (iv) the Board of Trustees of the Fund withdraws or materially modifies or changes its recommendation or approval of this Agreement in a manner adverse to Zenith or Zenith Insurance.

8.4 <u>Termination by the Fund</u>. This Agreement may be terminated by the Fund and the Merger may be abandoned at any time prior to the Closing Date, before or after the approval by Members of the Fund, (i) in the event of a breach by Zenith, Zenith Insurance or Zenith National of any covenant or agreement contained in this Agreement which, by its nature, cannot be cured prior to the Closing or which has not been cured within 30 days after the giving of written notice to Zenith of such breach, (ii) in the event of an inaccuracy of any representation or warranty of Zenith, Zenith Insurance or Zenith National contained in this Agreement which, by its nature, cannot be cured prior to the Closing or which has not been cured within 30 days after the giving of written notice to Zenith Insurance or Zenith National contained in this Agreement which, by its nature, cannot be cured prior to the Closing or which has not been cured within 30 days after the giving of written notice to Zenith of such inaccuracy and which inaccuracy, in either case, would cause the conditions set forth in Section 7.3(a) not to be satisfied, (iii) in the event that any of the conditions precedent to the obligations of the Fund to consummate the Merger cannot be satisfied or fulfilled by the date set forth in Section 8.2(ii) of this Agreement, provided that the failure of such conditions to be so satisfied shall not be as a result of the Fund's failure to fulfill its obligations under this Agreement, or (iv) prior to the Members Meeting, the Board of Trustees of the Fund has (y) withdrawn or modified or changed its recommendation or approval of this Agreement in a manner adverse to Zenith or Zenith Insurance in order to approve and permit the Fund to execute a definitive agreement relating to an Alternative Proposal and (z) determined, basis on the advice of outside legal counsel to the Fund, that the failure to take such action as set forth in the preceding clause (\vec{y}) would be reasonably likely to result in breach of the Board of Trustees' fiduciary duties under applicable Law; provided, however, that the Board of Trustees of the Fund shall have been advised by such outside counsel that notwithstanding a binding commitment to consummate transactions of the nature contemplated by this Agreement entered into in the proper exercise of their applicable fiduciary duties, such fiduciary duties would also be reasonably likely to require the trustees to terminate this Agreement as a result of such Alternative Proposal; provided, further, that the Fund shall immediately advise Zenith following the receipt by it of any Alternative Proposal and the details thereof, and advise Zenith of any developments with respect to such Alternative Proposal immediately upon the occurrence thereof.

8.5 <u>Effect of Termination</u>. In the event of termination of this Agreement and the abandonment of the Merger pursuant to this Article VIII, written notice thereof shall as promptly as practicable be given to the other party to this Agreement, and this Agreement shall terminate and the transactions contemplated hereby shall be abandoned, without further action by any of the parties hereto. If this Agreement is terminated as provided herein: (i) there shall be no liability or obligation on the part of Zenith, Zenith Insurance, Zenith National, the Fund, RMG or their respective trustees, officers and directors, and all obligations of the parties shall terminate, except (A) for the obligations of the parties pursuant to this Section 8.5, (B) for the provisions of Section 9.1, and (C) that a party who is in breach of any of its representations, warranties, coverants or agreements set forth in this Agreement shall be liable for damages occasioned by such breach, including, without limitation, any expenses incurred by the other party in connection with this Agreement, and (ii) all filings, applications and other submissions made pursuant to the transactions contemplated by this Agreement shall, to the extent practicable, be withdrawn from the agency or person to which it has been made.

ARTICLE IX

MISCELLANEOUS AND GENERAL

9.1 Payment of Expenses and Other Payments.

(a) Except as otherwise expressly provided herein, whether or not the Merger shall be consummated and except as otherwise provided in this Agreement, each party hereto shall pay its own expenses incident to preparing for, entering into and carrying out this Agreement and the consummation of the transactions contemplated hereby.

(b) The Fund shall pay, or cause to be paid, in same day funds to Zenith the sum of all Expenses upon demand, if (i) Zenith terminates this Agreement under Section 8.3(iv), (ii) the Fund terminates this Agreement under Section 8.4(iv), or (iii) prior to any termination of this Agreement, an Alternative Proposal shall have been made and within nine months of the termination of this Agreement a transaction constituting an Alternative Proposal is consummated with respect to, or the Fund enters into an agreement with respect to, or approves or recommends an Alternative Proposal, in each case made prior to any termination of this Agreement; provided, however, that in the case of (iii) above in this paragraph (b), no payment shall be made if this Agreement has been terminated pursuant to Sections 8.4(i) and (ii) hereof. The amount of Expenses so payable shall be the amount set forth in an estimate delivered by Zenith, subject to upward or downward adjustment (not to be in excess of the amount set forth in the definition of Expenses) upon delivery of reasonable documentation therefor.

9.2 <u>Survival of Representations and Warranties</u>. The respective representations and warranties of the parties made in Articles IV and V hereof shall not survive beyond the earlier of termination of this Agreement or the Effective Time.

9.3 <u>Modification or Amendment</u>. Subject to the applicable provisions of the FBCA, the GCL, the FIL and the California Insurance Code, at any time prior to the Effective Time, the parties hereto may modify or amend this Agreement, by written agreement executed and delivered by duly authorized officers of the respective parties; <u>provided</u>, <u>however</u>, that after approval of this Agreement by the Members of the Fund, no amendment shall be made which changes the consideration payable pursuant to the Merger or adversely affects the rights of the Fund's Members hereunder without the approval of such Members.

9.4 <u>Waiver and Extension</u>. At any time prior to the Effective Time, the parties hereto may (a) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (b) waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto or (c) except to the extent prohibited by Law, waive compliance with any of the agreements or conditions contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party. The failure of any party to this Agreement to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of those rights.

9.5 <u>Counterparts</u>. For the convenience of the parties hereto, this Agreement may be executed in any number of counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

9.6 <u>Governing Law</u>. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Florida without giving effect to the principles of conflicts of law thereof. Venue for the purposes of any cause of action, of whatsoever kind or nature, by any party against any other hereunder, shall be Orange County, Florida.

9.7 <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party to the other parties shall be in writing and shall be deemed given when delivered personally, upon receipt of a transmission confirmation (with a confirming copy sent by overnight courier) if sent by telecopy or like transmission, and on the next business day when sent by Federal Express, Express Mail, or other reputable overnight courier, as follows:

(a) If to the Fund or RMG, to

Doris Oberhardt Associated General Commerce Self-Insurers' Trust Fund 1363B Lafayette Street Tallahassee, Florida 32301 (904) 878-4261 (telephone) (904) 656-3237 (telecopier)

with a copy to:

Gerald S. Livingston, Esq. 255 South Orange Avenue Suite 850 Orlando, Florida 32801 (407) 422-2524 (telephone) (407) 422-2529 (telecopier)

If to Zenith, Zenith Insurance or Zenith National, to

John J. Tickner, Esq. Zenith Insurance Company 21255 Califa Street Woodland Hills, CA 91367-5021 (818) 594-5564 (telephone) (818) 594-7269 (telecopier)

with a copy to:

Skadden, Arps, Slate, Meagher & Flom 919 Third Avenue New York, New York 10019 Attention: Bertil Lundqvist, Esq. (212) 735-3000 (telephone) (212) 735-2000 (telecopier)

and

(b)

Skadden, Arps, Slate, Meagher & Flom 300 South Grand Avenue Los Angeles, CA 90071 Attention: Jerome L. Coben, Esq. (312) 687-5010 (telephone) (312) 687-5600 (telecopier)

or to such other persons or addresses as may be designated in writing by the party to receive such notice. Nothing in this Section 9.7 shall be deemed to constitute consent to the manner and address for service of process in connection with any legal proceeding (including litigation arising out of or in connection with this Agreement), which service shall be effected as required by applicable Law.

9.8 <u>Entire Agreement: Assignment</u>. This Agreement (a) constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes all other prior agreements and understandings, both written and oral, among the parties or any of them with respect to the subject matter hereof and (b) shall not be assigned by operation of law or otherwise.

9.9 <u>Parties in Interest</u>. This Agreement shall be binding upon and inure solely to the benefit of each party hereto and their respective successors and permitted assigns. Nothing in this Agreement, express or implied, other than the right to receive the consideration payable in the Merger pursuant to Article III hereof, is intended to cr shall confer upon any other person any rights, benefits or remedies of any nature whatsoever under or by reason of this Agreement; <u>provided</u>, <u>however</u>, that the provisions of Section 6.7 shall inure to the benefit of and be enforceable by the Indemnified Parties or Fund Lidemnified Parties, as the case may be.

9.10 Certain Definitions. As used herein:

"Accountant" shall have the meaning specified in Section 3.2.

"Adjusted GAAP Net Worth" shall have the meaning specified in Section 3.2.

"Alternative Proposal" shall mean (i) any proposal or offer for a merger, consolidation, asset acquisition, assumption reinsurance transaction, or other business combination involving the Fund or any proposal or offer to acquire a significant equity interest in, or a significant portion of the assets of, the Fund, other than the transactions contemplated by this Agreement, or (ii) any other transaction the consummation of which would reasonably be expected to impede, interfere with, prevent or materially delay the Merger.

"Articles of Merger" shall have the meaning specified in Section 1.2.

"Business of the Fund" shall mean workers' compensation insurance business written by the Fund prior to the Effective Time and acquired by Zenith.

"California Commissioner" shall mean the Commissioner of Insurance of the State of California.

0077359.10-21514

"Closing" shall mean the closing of the transactions contemplated by Article I of this

Agreement.

"Closing Balance Sheet" shall have the meaning specified in Section 3.2.

"Closing Date" shall mean the date on which the Closing occurs.

"Code" shall mean the Internal Revenue Code of 1986, as amended, and the rules and regulations promulgated thereunder.

"Commissioners" shall mean the California Commissioner and the Florida Commissioner.

"Consent" shall mean any consent, approval, authorization, clearance, exemption, waiver, or similar affirmation by, or filing with or notification to, a person pursuant to any Contract, Law, Order, or Permit.

"Consulting Agreement" shall have the meaning specified in Section 7.2(f).

"Contract" shall mean any written or oral agreement, arrangement, commitment, contract, indenture, instrument, lease or other obligation of any kind or character, or other obligation that is binding on any person or its capital stock, properties or business.

"Default" shall mean (i) any breach or violation of or default under any Contract, Order or Permit, (ii) any occurrence of any event that with the passage of time or the giving of notice or both would constitute a breach or violation of or default under any Contract, Order or Permit, or (iii) any occurrence of any event that with or without the passage of time or the giving of notice would give rise to a right to terminate or revoke, change the current terms of, or renegotiate, or to accelerate, increase, or impose any liability under, or create any Lien in Connection with, any Contract. Order or Permit.

"Disclosure Schedule" shall mean the disclosure schedule attached hereto.

"Effective Date" shall mean the date on which the Effective Time occurs.

"Effective Time" shall have the meaning assigned thereto in Section 1.2.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as

amended.

"ERISA Affiliate" shall mean any corporation or trade or business, whether or not incorporated, that together with an entity or any Subsidiary of such entity would be deemed a "single employer" within the meaning of Section 4001 of ERISA, or considered as being members of a controlled group of corporations, under common control, or members of an affiliated service group within the meaning of Subsections 414(b), (c), (m) or (o) of the Code or Section 4001(a)(14) of ERISA.

0077359.10-2151a

"ERISA Affiliate Plan" shall mean each employee benefit, welfare and compensation plan maintained, sponsored or contributed to be any ERISA Affiliate.

¢,

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

"Expenses" shall mean the out-of-pocket expenses of Zenith, Zenith Insurance and Zenith National, in an amount not to exceed \$150,000, payable in accordance with Section 9.1(b).

"FBCA" shall mean the Florida Business Corporations Act.

"FIL" shall mean the Florida Insurance Laws and the related rules and regulations thereunder as contained in the Florida Administrative Code, Rule Section 4-190.56 et. seq., F.A.C.

"Florida Commissioner" shall mean the Commissioner of Insurance of the State of

Florida.

"GAAP" shall mean generally accepted accounting principles, consistently applied throughout the specified period and in the immediately prior comparable period.

"GAAP Statements" shall mean the audited financial statements of the Fund as of, and for the fiscal years ended, December 31, 1993, 1994 and 1995, together with the notes related thereto.

"GCL" shall mean the California General Corporation Law.

"Governmental Authority" shall mean the government of the United States or any foreign country, any state or political subdivision thereof, or any entity, body or authority exercising executive, legislative, judicial, regulatory, administrative or other governmental functions or any court, department, commission, board, agency, instrumentality or administrative body of any of the foregoing.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indemnified Parties" shall mean the Fund's present or former trustees, officers, employees or agents covered by the Funds directors' and officers' liability insurance prior to the Effective Time.

"Interim Statements" shall mean the financial statements that present the financial position of the Fund as of June 30, 1996, and the results of its operations and cash flows for the six months then ended in accordance with GAAP.

"Law" shall mean any law, ordinance, regulation, rule, or statute of any governmental authority (Federal, state, local or otherwise) applicable to a person or its properties, liabilities or business.

٠.

"Liability for Assessments" shall mean the amount of contingent liability for which Members of the Fund are either jointly and severally liable, or jointly and proportionately liable, by reason of their participation as Members of the Fund.

"Lien" shall mean any conditional sale agreement, default of title, easement, encroachment, encumbrance, hypothecation, infringement, lien, mortgage, option, pledge, reservation, restriction, security interest, title retention or other security arrangement, or any adverse right or interest, charge, or claim of any nature whatsoever of, on, or with respect to any property or property interest.

"Litigation" shall mean any action, arbitration, cause of action, claim, complaint, criminal prosecution, demand letter, governmental or other administrative or other proceeding, whether at law or at equity, before or by any federal, state or foreign court, tribunal, or agency or before any arbitrator.

"Loss Sensitive Policy" shall mean a Retrospectively Rated Policy, including the policies written by the Fund known as Deluxe Retention Plan, Regular Retention Plan, Sliding Scale Plan and Retention Dividend Plan.

"Management Agreement" shall have the meaning set forth in Section 7.3(f).

"Management Group" shall mean, collectively, Howard Hice, John G. Martin, Ronald Morrick, Armand Mouw, Ronald Shafer, Jr., and Doris Oberhardt.

"Material Adverse Effect" shall mean any change or effect (or any development that, insofar as can reasonably be foreseen, is likely to result in any change or effect) that has a material adverse effect on (i) the business, properties, assets, financial condition, or results of operations or prospects of the Fund or RMG or (ii) the ability of the Fund or RMG to perform its obligations under this Agreement.

"Member(s)" or "Member(s) of the Fund" shall mean each employer member of the Fund who has executed an agreement to participate in the Fund and who has agreed to become bound by an indemnity agreement, which binds such member to individual, several, or proportionate liability as set forth in Sections 624.472 and 624.474 of the FIL, and as set forth in the indemnity agreement and/or amendments thereto filed with the Florida Department of Insurance, or previously filed with the Division of Workers Compensation, Florida Department of Labor and Employment Security, and who have further agreed to assume all obligations imposed upon them as set forth in the Florida Worker's Compensation Act, the rules and regulations adopted from time to time by the trustees of the Fund and the conditions of the application to the Department of Labor and Employment Security for membership in the Fund.

"Member Distributee" shall mean each corporation, partnership, person or other legal entity who was a Member of the Fund at any time during the five-year period ending on the date of the Effective Time.

"Members' Meeting" shall mean the special meeting of the Members contemplated by Section 6.3.

0077359.10-2151=

"Merger" shall have the meaning specified in Section 1.1.

"Merger Consideration" shall have the meaning specified in Section 3.1.

"Order" shall mean any administrative decision or award, decree, injunction, judgment, order, quasi-judicial decision or award, ruling, or writ of any federal, state, local or foreign or other court, arbitrator, mediator, tribunal, administrative agency or authority.

"PBGC" shall mean the Pension Benefit Guaranty Corporation.

"Permit" shall mean any federal, state, local or foreign governmental approval, authorization, certificate, declaration, easement, filing, franchise, license, notice, permit, variance, clearance, exemption, closure or right to which any person is a party or that is or may be binding upon or inure to the benefit of any person or its securities, properties or business.

"Plan of Distribution" shall mean the plan of distribution relating to the payment of the Merger consideration to the Member Distributees that has been approved by the Florida Department of Insurance.

[•]Plans[•] shall mean the employee benefit, welfare and compensation plans, programs, policies and arrangements maintained, sponsored or contributed to or required to be contributed to by the Fund or RMG.

"Proxy Statement" shall mean, collectively. " letter to policyholders, notice of meeting, proxy statement and form of proxy, or the information statement, as the case may be, to be distributed to Members in connection with the Merger, or any schedules required to be filed with the Florida Commissioner in connection therewith.

"Raymond James" shall mean Raymond James & Associates, Inc.

"Related Documents" shall mean the Consulting Agreement, the Management Agreement, the Transfer Agreement and the Proxy Statement.

"Requisite Percentage" shall mean 6635%.

"Reserve Liabilities" shall mean all reserves and other liabilities with respect to insurance and reinsurance and for claims and benefits incurred but not reported, as established or reflected in the SAP Statements.

"Retrospectively Rated Policy" shall mean a policy whose premium is established based on a review of policy experience subsequent to the expiration of the policy period.

"Revised Closing Balance Sheet" shall have the meaning specified in Section 3.2.

"RMG" shall mean AGC Risk Management Group, Inc., a not-for-profit corporation organized under the Business Corporation Law of the State of Florida.

0077359.10-21514

"SAP" shall mean the accounting practices required or permitted by the insurance regulatory authorities of the State of Florida, consistently applied throughout the specified period.

"SAP Statements" shall mean all of the Funds' annual and quarterly financial statements, together with all exhibits and schedules thereto, required to be filed with or submitted to the appropriate insurance regulatory authorities of the State of Florida on forms prescribed or permitted by such authorities utilizing such accounting methods as have been required from time to time by such authorities.

"September Financial Statements" shall have the meaning specified in Section 6.12.

"Surviving Corporation" shall mean Zenith, as the corporation surviving the Merger.

"Subsidiary" shall mean, when used with reference to any entity, any corporation with a majority of the outstanding voting securities owned directly or indirectly by such former entity.

"Tax Return" shall mean any report, return or other information required to be supplied to a Governmental Authority in connection with any Taxes.

"Taxes" shall mean all taxes, charges, fees, duties (including customs duties), levies or other assessments, including without limitation, income, gross receipts, net proceeds, ad valorem, turnover, real and personal property (tangible and intangible), sales, use, franchise, excise, value added, stamp, leasing, lease, user, transfer, fuel, excess profits, occupational, interest equalization, windfall profits, severance, license, payroll, environmental, capital stock, disability, employee's income withholding, other withholding, unemployment and Social Security taxes, which are imposed by an Governmental Authority, and such term shall include any interest, penalties or additions to tax attributable thereto.

"Transfer Agreement" shall have the meaning set forth in Section 7.2(g).

"Voting Member" shall mean each Member who (i) as reflected on the records of the Fund, was an owner of an in-force policy at the close of business on the record date for determining Members entitled to vote at the Members' Meeting, and (ii) casts a vote at the Members' Meeting.

"ZRM" shall mean a newly-formed Florida corporation, 100% of the stock of which will be owned by Zenith Insurance or Zenith National.

9.11 Validity. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provisions of this Agreement each of which shall remain in full force and effect.

9.12 <u>Captions</u>. The Article, Section and paragraph captions herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof.

9.13 <u>Further Assurances</u>. Each of the Fund and RMG agrees that, upon the reasonable request of Zenith, Zenith Insurance or Zenith National, it shall cooperate and cause its affiliates, employees and agents to cooperate with Zenith, Zenith Insurance and Zenith National to effect the orderly transition of the business, operations and affairs of the Fund and RMG to Zenith and ZRM, respectively. Without limiting the generality of the foregoing, RMG will (a) give, and cause its affiliates, employees and agents to give, Zenith, Zenith Insurance and Zenith National reasonable access to all books and records of the Fund and RMG, (b) cooperate with Zenith, Zenith Insurance and Zenith National in the preparation of any fluancial statements, reports or tax returns and in making any filings or claims for refund or responding to any audit or inquiry of any governmental or regulatory body, and (c) allow Zenith, Zenith Insurance and Zenith National to control (i) the preparation of any tax return or smended tax return to be filed by, or which includes, the Fund after the Closing Date, and (ii) any contest with any taxing authority concerning taxes of the Fund.

(The next page is the signature page)

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their respective duly authorized officers as of the date first above written. ••

Y

.

× *

Attest:	ZENI7 AGC ACQUISITION INSURANCE COMPANY
(seal)	-
	By:
	Name: Title:
Attest:	ZENITH INSURANCE COMPANY
[scal]	
	By:
	Name: Title:
Attest:	ZENITH NATIONAL INSURANCE CORP.
[scal]	
	Ву:
	Name: Title:
Attest:	ASSOCIATED GENELAL COMMERCE SELF INSURERS' TRUST FUND
[scal]	
	By:
	Name: Title:
Attest:	AGC RISK MANAGEMENT GROUP, INC.
[seal]	
م مرکز میں میں اور	By:Name:
	Tule:
0077359.10-21514	37

۰,

₹

Exhibit 7.2(c)

OPINION OF GERALD S. LIVINGSTON, ESQ., COUNSEL TO THE FUND

Gerald S. Livingston, Esq. will deliver an opinion, subject to customary assumptions, exceptions and conditions, to the effect that:

1. The Fund is a self-insurance trust fund duly organized, validly existing and in good standing under the Laws of the State of Florida. RMG is a not-for-profit corporation duly organized, validly existing and in good standing under the Laws of the State of Florida. Each of the Fund and RMG has all requisite power and authority and all necessary governmental Consents to own, lease and operate its properties and to carry on its business as it is now being to ducted. The Fund and RMG have heretofore made available to Zenith complete and correct copies of their Declaration of Trust and Animeles of Incorporation, respectively, and, in the case of RMG, it: By-Laws, each in effect as of the date hereof.

2. The Fund is (a) duly licensed or authorized as a self-insurance trust fund in the State of Florida, (b) not required to be licensed or authorized as a self-insurance trust fund or insurance company in any jurisdiction other than the State of Florida, and (c) duly authorized in the State of Florida to write the line of business reported as being written in the SAP Statements.

3. RMG is (a) duly licensed or authorized as a managing general agent and was previously licensed as a service company in the State of Florida, and (b) not required to be licensed or authorized as a managing general agent or service company in any jurisdiction other than the State of Florida.

4. The Fund has no Subsidiaries. Except as set forth in Section 4.2 of the Disclosure Schedule, the Fund does not directly or indirectly have any equity or similar interest in, or any interest convertible into or exchangeable or exercisable for any equity or similar interest in, any corporation, partnership, joint venture or other business association or entity that directly or indirectly conducts any activity which is material to the Fund. RMG has one member, which member is the Fund.

5. Each of the Fund and RMG has the requisite power and authority to execute and deliver the Agreement and to consummate the transactions contemplated thereby. The Agreement and the consummation by the Fund and RMG of the transactions contemplated thereby have been duly and validly authorized by the Board of Trustees of the Fund and the Board of Directors of RMG, and no other proceedings on the part of the Fund or RMG are necessary to authorize the Agreement or to consummate the transactions contemplated thereby. The Agreement has been duly and validly executed and delivered by the Fund and RMG and, assuming the Agreement constitutes the valid and binding agreement of Zenith, Zenith Insurance and Zenith National, constitutes the valid and binding agreement of the Fund and RMG, enforceable against the Fund and RMG in accordance with its terms, except that the enforcement hereof may be limited by (a) bankruptcy, insolvency, reorganization, moratorium or other similar Laws now or hereafter in effect relating to creditors' rights generally and (b) general principles of equity (regardless of whether enforceability is considered in a proceeding in equity or at law).

6. Neither the execution, delivery or performance of the Agreement by the Fund or RMG, nor the consummation by the Fund or RMG of the transactions contemplated thereby nor compliance by the Fund or RMG with any of the provisions thereof will (a) conflict with or result in any breach of any

provision of the D-claration of Trust of the Fund, or the Articles of Incorporation or By-Laws of RMG; (b) require any Consent of any governmental or regulatory authority, except the filing of Articles of Merger pursuant to the FBCA; (c) result in a Default under any of the terms, conditions or provisions of any Contract to which the Fund or RMG may be a party, or Permit by which the Fund or RMG or any of their ass-ts may be bound, except for such Defaults as to which requisite waives or consents have been obtained; or (d) violate any Order or Law applicable to the Fund or RMG or any of their assets.

7. Except as disclosed in the financial statements delivered pursuant to Section 4.5 of the Agreement or as set forth in Section 4.8 of the Disclosure Schedule, there is no Litigation pending or, to my knowledge, threatened against or affecting the business operations or financial condition of the Fund or RMG. Neither the Fund nor RMG is in default with respect to any judgement, order, writ, injunction or decree of any court or any Federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality.

2. The Proxy Statement complies with the FIL. None of the information as to the Fund or RMG contained in the Proxy Statement contains any untrue statement of a material fact required to be stated therein or omits to state a material fact necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

Such counsel's opinion shall be limited to the laws of the State of Florida and the federal laws of the United States of America, to the extent specifically referred to in such opinion.

Exhibit 7.2(1)

CONSULTING AGREEMENT

CONSULTING AGREEMENT (this "Agreement"), dated as of ______, 1996, by as _______, nong Howard Hice, John G. Martin, Ronald Morrick, Armand Mouw, Ron Shafer, Jr., and Doris Oberhards (hereinafter collectively referred to as the "Management Group") and ZENITH AGC ACQUISITION INSURANCE COMPANY, a Florida corporation (togs her with its successors and assigns, "Zenith").

<u>WITNESSETH</u>:

WHEREAS, Zenith, Zenith Insurance Company, Zenith National Insurance Corp., Associated General Commerce Self-Insurers' Trust Fund the "Fund") and AGC Risk Management Group, Inc. have heretofore entered into an Agreement and Plan of Merger, dated as of October 7, 1996 (as the same may be amended from time to time, the "Merger Agreement"), providing for, among other things, the merger of the Fund with and into Zenith (the "Merger");

WHEREAS, pursuant to Section 7.2(f) of the Merger Agreement, the obligation of Zenith to consummate the Merger is conditioned upon, among other things, the Management Group and Zenith entering into this Consulting Agreement; and

WHEREAS, Zenith desires to retain the Management Group's services as independent contractors and consultants in order to procure for itself the benefits of the extensive familiarity of the Management Group with the business and operations of the Fund, and the Management Group desires to make such services available to Zenith on the terms and conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants of the parties hereto and other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged and intending to be legally bound hereby, the parties hereto hereby agree as follows:

<u>FIRST</u>: The Management Group and Zenith agree that the Management Group will be retained as consultants for the period beginning at the Effective Time (as defined in the Merger Agreement) and ending on the fifth anniversary thereof (hereinafter referred to as the "Period"). The Management Group agrees to provide Zenith, its subsidiaries and affiliates with consulting services as requested by Zenith. Such consulting services shall be provided at mutually convenient times as Zenith may reasonably request during the Period. There will be no established schedule for the Management Group's services, and Zenith agrees to provide the Management Group with reasonable notice of the dates, times and places where the Management Group is or are to be available for consultation. Except as expressly provided herein, Zenith acknowledges that the services to be provided hereunder by the Management Group from engaging in any other full-time or part-time activity or employment in any business except for the Workers' Compensation Insurance Business (as hereinafter defined). The Management Group (except Doris Oberhardt) agrees that they shall not engage in or undertake any Workers' Compensation Insurance Business beginning on the date of this Agreement and for a period of five (5) years following the end of the Period. Notwithstanding anything to the contrary contained herein, it is expressly agreed and understood that Doris Oberhardt shall not engage

0077357,10-21514

in any other full-time or part-time activity or employment during the Period other than employment with Zenith Risk Management, Inc. I' is the expectation of the parties that the amount of time devoted by the Management Group (except Doris Oberhardt) to consulting services hereunder will decrease during the Period as Zenith becomes more familiar with the business and operations of the Fund. For purposes of this Agreement, "Workers' Compensation Insurance Business" shall mean the business of providing insurance coverage for income, medical, death, disability and/or rehabilitation benefits to employers for employee job-related injuries or diseases, and shall include any management, administration, or consulting services relating thereto.

SECOND: In consideration of the foregoing services, Zenith agrees to compensate each member of the Management Group in the amount of \$25,000 per year. Payment shall be made by Zenith within ten (10) days of the beginning of each payment period, except that payment for the first payment period will be made on the date of this Agreement.

<u>THIRD</u>: The Management Group agrees that, during the Period, the Management Group (except Doris Obe, hardt) will be independent contractors and therefore will be responsible for the payment of all taxes with respect to payments made under this Agreement. All payments under this Agreement will be made free and clear of, and without deduction or withholding for, any taxes. In the event that applicable law requires that any withholding be made in respect of the payment, to be made hereunder, Zenith agrees to withhold such amounts as may be required, pay such withheld amounts over to the applicable taxing authority and pay to the Management Group the remainder of the amount specified in paragraph SECOND hereof after deducting such withholding.

FOURTH: This Agreement may not be assigned by either party, other than by Zenith to a subsidiary or affiliate of Zenith, at any time during the Period without the prior written consent of the other party.

FIFTH: Any notice, request, instruction or other document to be given hereunder by any party to the other parties shall be in writing and shall be deemed given when delivered personally, upon receipt of a transmission confirmation (with a confirming copy sent by overnight courier) if sent by telecopy or like transmission, and on the next business day when sent by Federal Express, Express Mail, or other reputable overnight courier as follows:

(a) If to the Management Group, to:

Howard Hice [insert address, telephone, telecopy]

John G. Martin (insert address, telephone, telecopy)

Ronald Morrick [insert address, telephone, telecopy]

Armand Mouw [insert address, telephone, telecopy]

0077339,10-2151+

Ron Shafer, Jr. [insert address, telephone, telecopy]

and

Doris Oberhardt [insert address, telephone, telecopy]

with a copy to:

Gerald S. Livingston, Esq. 255 South Orange Avenue Suite 850 Orlando, Florida 32801 (407) 422-2524 (telephone) (407) 422-2529 (telecopier)

(b) If to Zenith, to it at:

Zenith AGC Acquisition Insurance Company c/o Zenith Insurance Company 21255 Califa Street Woodland Hills, CA 91367-5021 Attention: John J. Tickner, Esq. Telephone: (818) 594-5564 Telecopy: (818) 594-7269

with a copy to:

Bertil Lundqvist, Esq. Skadden, Arps, Slate, Meagher & Flom 919 Third Avenue New York, NY 10028 Telephone: (212) 735-3000 Telecopy: (212) 735-2000

and

Jerome L. Coben, Esq. Skadden, Arps, Slate, Meagher & Flom 300 South Grand Avenue Los Angeles, CA 90071 Telephone: (312) 687-5010 Telecopy: (312) 687-5600

0077359.10-2151+

1

÷

1

or to such other persons or addresses as may be designated in writing by the party to receive such notice. Nothing in this Section FIFTH shall be deemed to constitute consent to the manner and address for service of process in connection with any legal proceeding (including litigation arising out of or in connection with this Agreement), which service shall be effected as required by applicable law.

SIXTH: This Agreement will be governed by, and construed in accordance with, the laws of the State of Florida without giving effect to the principles of conflicts of law thereof. Venue for the purposes of any cause of action, of whatsoever kind or nature, by any party against any other hereunder, shall be Orange County, Florida.

SEVENTH: This Agreement constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes all other prior agreements and understandings, both written and oral, among the parties or any of them with respect to the subject matter hereof.

EIGHTH: All information disclosed by any party to the other party in connection with this Agreement or the services to be provided hereunder shall be kept confidential by the party receiving such information and no public statements shall be issued by any party relating to such information without the other party's prior written consent. The obligations contained in this Paragraph EIGHTH shall survive the termination of this Agreement.

<u>NINTH</u>: This Agreement shall be binding upon and inure solely to the benefit of each party hereto and their respective successors and permitted assigns. Nothing in this Agreement, express or implied, is intended to or shall confer upon any other person any rights, benefits or remedies of any nature whatsoever under or by reason of this Agreement.

<u>TENTH</u>: For the convenience of the parties hereto, this Agreement may be executed in any number of counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

ELEVENTH: The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement each of which shall remain in full force and effect.

<u>TWELFTH</u>: Zenith agrees to indemnify and hold harmless the Management Group from and against all costs, damages, judgments, attorney's fees, expenses, obligations and liabilities of whatsoever kind or nature, which, in good faith, the Management Group may incur or sustain in connection with or arising from the Management Group's performance hereunder, except such costs, damages, judgments, attorney's fees, expenses, obligations or liabilities which may be incurred by the Management Group as a result of (a) a violation of a criminal law resulting in a conviction, (b) any deliberately dishonest or fraudulent act or omission, or (c) any act or omission committed in bad faith or with malicious purpose or constituting gross negligence.

<u>THIRTEENTH</u>: This Agreement shall terminate in the event that the Merger Agreement is terminated, in which case the parties hereto shall have no liability to each other under this Agreement (except as expressly provided in the Merger Agreement). In the event that any member of the Management Group shall die or become incapacitated such that he or she is unable to perform his or her duties under this Agreement, then Zenith's obligation to compensate such member of the Management Group pursuant to paragraph SECOND shall terminate.

••

2

Ο

ł

٠.

(the next page is the signature page)

.

•

."

•

;

IN WITNESS WHEREOF, the parties hereto have each executed this Abreement as of the date first written above.

ZENITH AGC ACQUISITION INSULANCE COMPANY

...

b

By:___

Name: Title:

Howard Hice

John G. Martin

Ronald Morrick

Armand Mouw

Ronald Shafer, Jr.

Deris Oberhardt

0077359.10-21514

0

.

Exhibit 7.2(g)

TRANSFER AGREEMENT

TRANSFER AGREEMENT (this "Agreement"), dated as of ______, 1996, by and between AGC RISK MANAGEMENT GROUP, INC., a Florida not-for-profit corporation ("RMG"), and ZENITH RISK MANAGEMENT, INC., a Florida corporation ("ZRM"). Terms used but not otherwise defined herein shall have the respective meanings ascribed to them in the Agreement and Plan of Merger dated as of October 7, 1996, by and among Zenith AGC Acquisition Insurance Company, a Florida corporation ("Zenith"), Zenith Insurance Company, a California corporation ("Zenith Insurance"), Zenith National Insurance Corp., a Delaware corporation ("Zenith National"), Associated General Commerce Self-Insurers' Trust Fund, a qualified Florida self-insurers fund (the "Fund"), and AGC Risk Management Group, Inc., a Florida not-for-profit corporation ("RMG") (as the same may be amended from time to time, the "Merger Agreement").

<u>WITNESSETH</u>:

WHEREAS, Zenith, Zenith Insurance, Zenith National, the Fund and RMG have heretofore entered into the Merger Agreement providing for the merger of the Fund with and into Zenith pursuant to the terms and conditions of the Merger Agreement (the "Merger"), and the subsequent merger of Zenith with and into Zenith Insurance; and

WHEREAS, pursuant to section 7.2(g) of the Merger Agreement, the obligation of Zenith to consummate the Merger is conditioned upon, among other things, the execution and delivery by RMG and ZRM of this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants of the parties hereto and other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

L. As of the Effective Time, RMG hereby assigns, sells, transfers, conveys and delivers to ZRM, its successors and assigns all of RMG's right, title and interest in and to all of RMG's assets, including the assets identified in <u>Schedule A</u> hereto (collectively, the "Assets").

2. As of the Effective Time, ZRM hereby purchases and accepts the assignment and transfer of all of RMG's right, title and interest in and to the Assets, and hereby assumes the liabilities and obligations of PMG set forth on <u>Schedule B</u> hereto.

3. RMG hereby represents and warrants that (a) the Assets constitute all of the assets currently being used by RMG in its business, (b) RMG has good and marketable title to the Assets, with the free and unencumbered right to transfer the same, and that the Assets are free and clear of all liens and encumbrances, and (c) ZRM and its successors and assigns will have all right, title and interest in and to the Assets upon the delivery of this Agreement.

4. RMG hereby agrees and acknowledges that ZRM may hire any employees of RMG that ZRM deems to be necessary or useful in the operation or management of ZRM's business.

0077359.10-21514

RMG waives any and all claims it may have against ZRM or any former employee of RMG which may be hired by ZRM, whether such claims are based on written employment agreements with such employees or otherwise. RMG's waiver of its rights pursuant to this paragraph shall be deemed to be for the benefit of both ZRM and a y such employee. KMG shall take any and all actions, consistent with applicable law, as ZRM may request with respect to the Plans, including but not limited to actions concerning the termination of any Plan and/or the transfer, distribution and/or other disposition of the assets of any Plan.

5. As soon as practical after the Effective Time, RMG shall take all appropriate action to effect its dissolution and liquidation in accordance with Florida law.

6. For the convenience of the parties hereto, this Agreement may be executed in any number of counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.

7. This Agreement shall be governed by, and construed in accordance with the laws of the State for Florida without giving effect to the principles of conflicts of law thereof. Venue for the purposes of any cause of action, of whatsoever kind or nature by any party against any other hereunder, shall be Orange County, Florida.

8. RMG an i ZRM, as reasonably requested to do so by the other party from time to time, shall, at their own expense, do, execute, acknowledge and deliver any and all such other further acts, transfers and any instruments or further assurances, approvals and consents as are necessary or proper in order to accomplish and complete the transactions contemplated hereby and the purposes hereof.

9. Any notice, request, instruction or other document to be given hereunder by any party to the other parties shall be in writing and shall be deemed given when delivered personally, upon receipt of the transmission confirmation (with a confirming copy sent by overnight courier) if sent by telecopy or like transmission, and on the next business day when sent by Federal Express, Express Mail, or other reputable overnight courier, as follows:

(a) If to RMG, to:

Doris Oberhardt Associated General Commerce Self-Insurers' Trust Fund 1363 East Lafayette Street Tallahassee, Florida 32301 Telephone: (904) 878-4261 Telecopy: (904) 656-3237

0077359.10-2151a

with a copy to:

Gerald S. Livingston, Esq. 255 South Orange Avenue Suite 850 Orlando, Florida 32801 Telephone: (407) 422-2524 Telecopy: (407) 422-2529

(b) If to ZRM, to:

John J. Tickner, Esq. Zenith Risk Management, Inc. c/o Zenith Insurance Company 21255 Califa Street Woodland Hills, California 91367 Tclephone: (818) 594-5564 Telecopy: (818) 594-7269

with a copy to:

Bertil Lundqvist, Esq. Skadden, Arps, Slate, Meagher & Flom 919 Third Avenue New York, NY 10028 Telephone: (212) 735-3000 Telecopy: (212) 735-2000

and

1

Jerome L. Coben, Esq. Skadden, Arps, Slate, Meagher & Flom 300 South Grand Avenue, Suite 3400 Los Angeles, California 90071 Telephone: (213) 687-5010 Telecopy: (213) 687-5600

or to such other persons or addresses as may be designated in writing by the party to receive such notice. Nothing in this paragraph 9 shall be deemed to constitute consent to the manner and address for service of process in connection with any legal proceeding (including litigation arising out of or in connection with this Agreement), which service shall be effected as required by applicable law.

10 This Agreement (a) constitutes the entire agreement among the parti s with respect to the subject matter hereof and supersedes all other prior agreements and understanding, both written and oral, among the parties or any of them with respect to the subject matter hereof and (t) shall not be assigned by operation of law or otherwise.